JGR Space Physics

RESEARCH ARTICLE

10.1029/2021JA030115

Special Section:

Geospace multi-point observations in Van Allen Probes and Arase era

Key Points:

- Symmetric compressional ULF (4–10 mHz) waves were excited at L = 3.0–5.5 during a geomagnetic storm
- The waves were associated with enhancement of the flux of protons at energy <5 keV
- Drift compressional instability is a possible wave generation mechanism

Correspondence to:

K. Takahashi, kazue.takahashi@jhuapl.edu

Citation:

Takahashi, K., Crabtree, C., Ukhorskiy, A. Y., Boyd, A., Denton, R. E., Turner, D., et al. (2022). Van Allen Probes observations of symmetric stormtime compressional ULF waves. *Journal of Geophysical Research: Space Physics*, *127*, e2021JA030115. https://doi. org/10.1029/2021JA030115

Received 8 NOV 2021 Accepted 23 JAN 2022

Author Contributions:

Conceptualization: Kazue Takahashi Data curation: Kazue Takahashi, Massimo Vellante Formal analysis: Kazue Takahashi, Chris Crabtree, Massimo Vellante Funding acquisition: Kazue Takahashi, Chris Crabtree, A. Y. Ukhorskiy Investigation: Kazue Takahashi, Chris Crabtree, A. Y. Ukhorskiy Methodology: Kazue Takahashi, Alexander Boyd Project Administration: Kazue Takahashi, A. Y. Ukhorskiy, Harlan E. Spence Resources: Kazue Takahashi

© 2022 The Johns Hopkins University, Applied Physics Laboratory. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Van Allen Probes Observations of Symmetric Stormtime Compressional ULF Waves

Kazue Takahashi¹, Chris Crabtree², A. Y. Ukhorskiy¹, Alexander Boyd³, Richard E. Denton⁴, D. Drew Turner¹, Matina Gkioulidou¹, Massimo Vellante^{5,6}, and Harlan E. Spence⁷

¹The Johns Hopkins University Applied Physics Laboratory, Laurel, MD, USA, ²Naval Research Laboratory, Washington, DC, USA, ³Department of Space Science, Aerospace Corporation, Chantilly, VA, USA, ⁴Department of Physics and Astronomy, Dartmouth College, Hanover, NH, USA, ⁵Department of Physical and Chemical Sciences, University of L'Aquila, L'Aquila, Italy, ⁶Consorzio Area di Ricerca in Astrogeofisica, L'Aquila, Italy, ⁷Institute for the Study of Earth, Oceans, and Space, University of New Hampshire, Durham, NH, USA

Abstract Previous spacecraft studies showed that stormtime poloidal ultralow-frequency (ULF) waves in the ring current region have an antisymmetric (second harmonic) mode structure about the magnetic equator. This paper reports Van Allen Probes observations of symmetric ULF waves in the postnoon sector during a moderate geomagnetic storm. The mode structure is determined from the presence of purely compressional magnetic field oscillations at the equator accompanied by strong transverse electric field perturbations. Antisymmetric waves were also detected but only very late in the recovery phase. The symmetric waves were detected outside the plasmasphere at L = 3.0-5.5 and had peak power at 4–10 mHz, lower than the frequency of the local fundamental toroidal standing Alfvén wave. During the wave events, the flux of protons was enhanced at energies below ~5 keV, which appears to be a prerequisite for the waves. The protons may provide free energies to waves through drift resonance instability or drift compressional instability, which occur in the presence of radial gradients of plasma parameters.

Plain Language Summary During geomagnetic storms, the intensity of the flux of ions is elevated in the magnetosphere including the region inward of geostationary orbits. The ions are capable of exciting a variety of plasma waves including ultralow-frequency (ULF) waves in the frequency range 1–20 mHz. Stormtime ULF waves with a strong perturbation of the magnetic field magnitude are commonly detected at geostationary orbit and are known to commonly have a standing wave structure along the background magnetic field that is antisymmetric about the magnetic equator. Observations made by the Van Allen Probes provide new evidence that symmetric waves are also excited inward of geostationary orbits. The waves may be excited by the spatial gradient of parameters associated with the ions.

1. Introduction

Ultralow-frequency (ULF) waves in the Pc4–5 band (period = 45–600 s) with strong radial and compressional magnetic field components (compressional poloidal waves) are observed at L > 5 within the ring current region (L = 3-8) during geomagnetic storms (Anderson et al., 1990; Barfield & McPherron, 1972; Lanzerotti et al., 1969; Le et al., 2017; Sonnerup et al., 1969; Takahashi et al., 1985). Similar waves are observed at large distances (L > 8) even when the geomagnetic activity is low (Constantinescu et al., 2009; Nishida et al., 1997; Takahashi et al., 1990; Vaivads et al., 2001; Zhu & Kivelson, 1991). Although these waves are believed to be excited by ion-driven instabilities (e.g., Chen & Hasegawa, 1991; Southwood, 1976), they can also interact with electrons. For example, the waves may heat electrons (Lanzerotti et al., 1969), transport them radially (Ukhorskiy et al., 2009), or drive them into the atmospheric loss cone (Rae et al., 2018).

An important factor in studying the excitation of compressional Pc4–5 waves and the interaction of the waves with charged particles is the mode structure of the waves along the background magnetic field (e.g., Southwood, 1976). ULF waves are subjected to the ionospheric boundary condition. As a consequence, the waves commonly establish a standing wave structure along the background magnetic field. The structure dictates what type of resonance is possible between the waves and particles that are executing drift and bounce motion. Figure 1 illustrates the mode structure of standing waves, using a locally defined orthogonal magnetic field aligned (MFA)







Software: Kazue Takahashi, A. Y.coordinate system :Ukhorskiy, Alexander Boyd ϕ (eastward), and μ Validation: Kazue Takahashi, Matinaand magnetic fieldGkioulidouVisualization: Kazue Takahashi

Writing – original draft: Kazue Takahashi

Writing – review & editing: Chris Crabtree, Alexander Boyd, Richard E. Denton, Drew Turner, Matina Gkioulidou, Massimo Vellante, Harlan E. Spence coordinate system for the wave-induced perturbations. The three axes in the MFA system are labeled ν (outward), ϕ (eastward), and μ (field aligned). We use the symbols ξ , **E**, and **B** for the field line displacement, electric field, and magnetic field, respectively.

Assuming that the background **B**-field and plasma parameters are symmetric about the magnetic equator, we can classify the standing waves into symmetric (Figures 1b and 1c) and antisymmetric (Figures 1d and 1e) modes in terms of the pattern of ξ about the magnetic equator. The observable field components, E_{ν} , E_{ϕ} , and B_{μ} have the same symmetry as ξ , whereas B_{ν} and B_{ϕ} have the opposite symmetry. In general, waves with lower mode numbers carry higher energies, so the highest wave power is usually found in the fundamental (symmetric) or second (antisymmetric) harmonic mode. The fundamental and second harmonic poloidal waves, denoted P1 and P2 waves, are examples of the symmetric and antisymmetric modes observed in the magnetosphere. The illustration indicates that it is straightforward to determine the symmetry of a standing wave with a spacecraft located at the equator. If the wave has a symmetric structure, the spacecraft will detect perturbations in E_{ν} , E_{ϕ} , and B_{μ} (antinode) but not in B_{ν} or B_{ϕ} (node). If the wave has an antisymmetric structure, the detectable and undetectable field components are switched.

Theory indicates that the transverse and compressional components of ULF waves are generally coupled. In magnetohydrodynamic (MHD) wave theory, the inhomogeneity of the magnetosphere couples the transverse (Alfvén) waves and the compressional (fast mode) waves except at the limit of m = 0 (Radoski & Carovillano, 1966) or $|m| = \infty$ (Radoski, 1967), where *m* is the azimuthal wave number defined positive (negative) for eastward (westward) propagation. In a kinetic approach, Crabtree and Chen (2004) showed that the compressional component is coupled to the shear Alfvén component, as found in our case, except when the frequency is well below the shear Alfvén frequency.

Previous studies have shown that instabilities in the ring current preferentially excite antisymmetric waves. Studies of stormtime Pc5 waves observed by geostationary satellites indicated that the waves had a second harmonic (antisymmetric) structure (Le et al., 2017; Takahashi et al., 1987). Antisymmetric waves are excited during quiet times as well (W. Liu et al., 2013). These observations have been explained by mechanisms such as the drift bounce resonance instability (Southwood, 1976) and drift Alfvén ballooning instability (Chen & Hasegawa, 1991).

Symmetric poloidal standing waves do occur in the ring current region, but observations of these waves are mostly limited to quiet times. The most extensively studied symmetric waves are P1 waves (Dai et al., 2015; Motoba et al., 2015; Takahashi et al., 2011; Takahashi, Claudepierre, et al., 2018; Yamamoto et al., 2018). P1 waves exhibit sinusoidal waveforms and are considered to be excited through drift resonance of energetic (\sim 100 keV) ions (Thompson & Kivelson, 2001) in the presence of an inward gradient of the phase space density of the resonant ions (Dai et al., 2015; Takahashi, Claudepierre, et al., 2018; Yamamoto et al., 2018). Similar symmetric poloidal waves were detected by the Cluster spacecraft, but their association with ground pulsation was not discussed, and no conclusion was drawn on how the waves were excited (Eriksson et al., 2005, 2006). P1 waves are the source of giant pulsations (Pgs) observed on the ground (Takahashi et al., 2011; Takahashi, Claudepierre, et al., 2018; Yamamoto et al., 2018). Pgs are observed at $L \sim$ 6 over a wide range of magnetic local time (MLT; 01–18 hr) at times of low geomagnetic activity (Brekke et al., 1987; Motoba et al., 2015).

This study reports observation of symmetric stormtime compressional Pc4–5 waves by Van Allen Probes (Radiation Belt Storm Probes, RBSP) in the dayside magnetosphere. These waves have peak power at L < 5 and exhibit more irregular waveforms than quiet-time P1 waves. Being excited in a region of strong magnetic field, the ion pressure is not high enough to drive the waves through the drift mirror instability (DMI). We suggest that the waves are instead driven by a gradient-driven instability such as drift resonance instability (DRI) or drift compressional instability (DCI).

The remainder of this study is as follows. Section 2 describes experiments and data. Section 3 presents an overview of wave observations. Section 4 describes the properties of the waves. Section 5 presents a brief review of possible wave excitation mechanisms. Section 6 presents the instability analysis. Section 7 examines the electron response to the waves. Section 8 presents the discussion, and Section 9 presents the conclusions.





Figure 1. Illustration of standing waves on the background magnetic field. (a) Coordinate system. (b) Structure of the ξ_{ν} , ξ_{ϕ} , E_{ν} , E_{ϕ} , and B_{μ} components of symmetric waves. The fundamental mode is considered. (c) Structure of the B_{ν} and B_{ϕ} components of symmetric waves. Fundamental toroidal (T1) and poloidal (P1) waves have this structure. (d, e) same as (b, c) but for antisymmetric waves. The second harmonic wave is considered. Second harmonic poloidal (P2) waves have this structure.

2. Experiments and Data

Data used in this study mainly come from the RBSP A and B spacecraft (Mauk et al., 2012). The data include **E**-field (Wygant et al., 2013); **B**-field (Kletzing et al., 2013); electron density (n_e) determined from plasma wave spectra (Kurth et al., 2015); ion differential energy flux measured at 0.985 eV to 51.8 keV by the Helium, Oxygen, Proton, and Electron (HOPE) mass spectrometer (Funsten et al., 2013) and at 44.7–598 keV by the Radiation Belt Storm Probes Ion Composition Experiment (RBSPICE; Mitchell et al., 2013), and electron differential energy flux measured at 33 keV to 4.1 MeV by the Magnetic Electron Ion Spectrometer (MagEIS; Blake et al., 2013). We exclude RBSPICE data at the lowest energy 44.7 keV because of a high noise level. We use **E**-field data that are obtained by first applying the spinfit method to the two components measured in the spacecraft spin plane and then by using the **E**·**B** = 0 assumption to derive the spin axis component (Wygant et al., 2013). The data have a time resolution of the spacecraft spin period (~11 s). As for the magnetic field, we use 1-s samples or its smoothed version (11-s running averages) sampled at the time stamps of the **E**-field data.

We rotate the **E** and **B** vectors into MFA coordinates (Figure 1a). For the **E**-field, the μ axis is the direction of the measured magnetic field averaged over the spin period. For the **B**-field, we use two versions of the MFA system. One uses the T89c magnetic field model (Tsyganenko, 1989) to define the μ axis. This model is more realistic than the dipole field but does not incorporate the ring current effect in any specific way. We intentionally use the model to see the effects of the ring current on the background magnetic field, as we discuss in Section 4.1. The other, used for spectral analysis, defines the μ axis for a short (e.g., 15 min) data segment by the trend magnetic field **B**_{trend} that is obtained by fitting a polynomial to the three components of the observed magnetic field (Takahashi & Denton, 2021; Zhu & Kivelson, 1994), with the degree of the polynomial set to four. In this system, the B_{ν} and B_{ϕ} components are transverse perturbations about **B**_{trend}, and the compressional perturbation is given by $B_{\mu} - |\mathbf{B}_{trend}|$. For wave analysis, the **E**-field data are also detrended by subtraction of a polynomial trend function from each component.

The spacecraft data are supplemented by data from the European Quasi-Meridional Magnetometer Array (EMMA; Lichtenberger et al., 2013). During the time interval studied, 24 EMMA magnetometers were in operation spanning L = 1.57-6.49. We use the standard *H* (horizontal northward) and *D* (horizontal eastward) coordinate system for the ground magnetic field data. We also use solar wind OMNI data and geomagnetic activity indices AU, AL, and Dst.



3. Overview of Wave Observation

Figure 2 presents an overview of the space environment and magnetospheric ULF waves on August 26–30, 2015. The top three panels show solar wind parameters taken from the OMNI data provided in 1 min resolution: the ion bulk velocity V_i , the dynamic pressure P_{dyn} , and the *z* component of the magnetic field in geocentric solar magnetospheric coordinates B_{zGSM} . The velocity is moderate at 400–500 km/s and shows smooth and slow variations. In contrast, P_{dyn} varies significantly between 0.9 and 9.2 nPa. Most importantly, B_{zGSM} varies between -16 and 10 nT with negative values dominating from midday on August 26 until the end of August 28. This B_{zGSM} behavior explains the ~2 days interval spanning August 26–29 of repeated enhancements of the auroral electrojet according to the AL and AU indices (Figure 2d) and a moderately developed ring current according to the Dst values between -100 and -40 nT (Figure 2f). The Dst index does not follow a simple pattern of a storm: a single main phase (Dst decrease) and a single recovery phase (Dst increase). Instead, we see quasiperiodic variations with minima occurring ~12 hr apart.

Figures 2g–2i show the locations of RBSP A (red) and RBSP B (blue) in dipole coordinates: geocentric equatorial distance of the field line that passes the spacecraft (*L*), magnetic latitude (MLAT, in degrees), and MLT (in hr). The thick line segments in these figures indicate the time intervals of poloidal ULF waves in the Pc4–5 band (1.7–22 mHz) that are identified by visual inspection of time series plots and dynamic spectra of the **E** and **B** fields measured by the spacecraft. By "poloidal," we mean substantial perturbations in at least two of the E_{ϕ} , B_{ν} , and B_{μ} components. On some orbits, the wave detection occurred very close to the magnetic equator (|MALT| < 3°), which enables us to distinguish between symmetric and antisymmetric waves from their node and antinode signatures at the equator (Figure 1). The shading in Figures 2a–2i highlights four such equatorial observations with RBSP A, which occurred on the outbound leg of orbits 2920, 2922, 2925, and 2928. On the first three of these orbits, symmetric waves (shaded yellow) were detected.

We determined the symmetry of the waves using the magnetic field power spectral density (PSD) shown in Figures 2j-2m, plotted as a function of *L*. We include n_e at the top and indicate the location of the outer edge of the electron plasmapause (L_{pp}) by an arrow directed upward. On the first orbit (2920), plasmapause crossing occurred at L = 2.5. On the second orbit (2922), the plasmapause was apparently located at L < 2.1. On the third orbit (2925), the crossing occurred at L = 2.9. By the time of the fourth orbit (2928), the plasmapause either had been smoothed out or had moved beyond spacecraft apogee (L = 5.8).

Included in each spectrogram is the model fundamental toroidal (T1) wave frequency labeled $f_{T1_Cummings}$. We obtained $f_{T1_Cummings}$ by solving the toroidal wave equation of Cummings et al. (1969) for a dipole magnetic field assuming that the mass density (ρ) varies along the magnetic field line as 1/r, where *r* is geocentric distance. Use of the dipole field is justified because we are interested in the region inward of L = 5, where the dipole field is strong, and the geomagnetic storm was not very strong according to the Dst values shown in Figure 2f. Following previous magnetoseismic studies (e.g., Takahashi et al., 2006), we introduce the average ion mass M_i ($\equiv \rho/n_e$) and express the mass density as $\rho = n_e M_i$. We set M_i at 3 amu following the statistical result obtained using data from the Combined Release and Radiation Effects Satellite (CRRES; Takahashi et al., 2006). We adopted the same M_i value for all orbits shown in Figure 2.

The B_{ν} and B_{μ} PSD shown in the second and third panels of Figures 2j–2m demonstrates changes in wave properties through the course of the storm. The spectrograms for the first three orbits show strong B_{μ} power peaked at 4–10 mHz within L = 3-5.5. This frequency is well below $f_{T1_Cummings}$. There is no corresponding power enhancement in the B_{ν} component. We take the absence of B_{ν} power as evidence for an equatorial node of that component because the spacecraft was virtually on the equator during these wave events. We conclude that symmetric waves were detected on the first three orbits because symmetric waves have an equatorial node of B_{ν} and an equatorial antinode of B_{μ} .

We see different waves on the fourth orbit (Figure 2m). The waves appear at L = 4.1-5.8 only in the B_{ν} spectrogram, and their frequency is $\sim 2f_{T1}$, falling steadily as L increases. The strong B_{ν} component and the absence of B_{μ} perturbation indicate that the waves had an antisymmetric structure. Both the frequency and the





Figure 2.



polarization indicate that the waves are P2 waves, which are commonly excited in the ring current region (Cummings et al., 1969). Recent RBSP studies (Min et al., 2017; Oimatsu et al., 2018; Takahashi, Oimatsu, et al., 2018; Yamamoto et al., 2019) presented detailed analysis of the physical properties and the generation mechanism of P2 waves. Z. Y. Liu et al. (2020) studied the ion response to the P2 waves reported by Yamamoto et al. (2019).

4. Wave Properties

We examine the symmetric compressional waves in detail, focusing on observations made on August 27, 2015 by RBSP A (orbit 2920), RBSP B (orbit 2904), and EMMA. Figure 3a shows the locations of the spacecraft and the EMMA magnetometers at 0830–1230 UT in *L*-MLT polar coordinates. The two spacecraft had nearly identical orbits, with RBSP B trailing RBSP A by ~70 min. The EMMA stations covered the *L* and MLT positions of the spacecraft very well. The heavy portions of the spacecraft orbit traces indicate where the symmetric waves were detected.

4.1. Localization in L

The waves are localized in *L*. Figures 3b-3d compare the magnetic field data from RBSP A and RBSP B. The MFA coordinate system for this figure is based on the T89c model (Tsyganenko, 1989), with Figure 3d showing the difference between the B_{μ} component and the magnitude of the T89c field. The median values of the field components are removed, and ± 10 nT offsets are added to separate data from the two spacecraft. The spacecraft positions in *L* and MLT are also plotted (Figures 3e and 3f). The magnetic field data indicate that the waves of interest to us produce only compressional (B_{μ}) perturbations.

Figure 3d indicates a time delay of the compressional wave event between the two spacecraft. RBSP A detected the waves at 0902–1026 UT, which corresponds to the *L* range 3.8–5.3 (Figure 3e, heavy line). RBSP B detected the event ~1 hr later at 1012–1128 UT, but in a very similar *L* range, 3.7–5.0. During the time interval of overlap (1012–1026 UT), when the spacecraft were separated ~1.3 in *L* and ~0.2 hr in MLT, the waveform differs between the two spacecraft. These features imply that the compressional waves are localized in *L* and had a coherence length shorter than $1.3 R_{\rm E}$.

The compressional oscillations are detected in a region where the observed B_{μ} is depressed relative to the T89c model. The depression, which has a magnitude of ~20 nT, appears because the model does not represent the effects of storm ring current. The collocated wave activity and magnetic field depression imply that the waves were excited by an instability associated with elevated plasma pressure.

4.2. Frequency and Polarization

Figure 4 shows frequency versus *L* spectrograms for all field components at RBSP B on the outbound leg of orbit 2904. The B_{ν} and B_{μ} spectrograms are very similar to those at RBSP A (Figure 2j) regarding both frequency (~7 mHz) and localization in *L* (3.5–5.0). Both E_{ν} and E_{ϕ} exhibit power spectra very similar to those of B_{μ} . The strong **E**-field power is what we expect for symmetric waves. There is only weak power in the B_{ϕ} component consistent with symmetric waves observed near the magnetic equator (|MLAT| < 3°).

To address the question of whether the compressional waves are strongly coupled to standing Alfvén waves (e.g., Chen & Hasegawa, 1991), we estimated the frequency of fundamental toroidal waves (denoted f_{TI}) by applying the cross-phase analysis technique (Waters et al., 1991) to EMMA data for 0900–1000 UT. We chose this time interval because the EMMA stations were close to 12 hr MLT and the cross-phase signatures are clear at many

Figure 2. Overview of observation of ULF waves by RBSP A and B on August 26–30, 2015. (a–c) Solar wind parameters selected from the OMNI data. (d) Magnetopause standoff distance according to Shue et al. (1998). The horizontal dashed line indicates the maximum *L* (6.5) reached by the RBSP spacecraft. (e, f) Geomagnetic activity indices. (g–i) Dipole coordinates of RBSP A (red) and B (blue). The colored thick portion of the coordinate plots indicates detection of poloidal Pc4–5 waves. The yellow or gray shading highlights observations by RBSP A near the magnetic equator on its outbound orbit legs. (j–m) Electron number density and the power spectra of the B_{ν} and B_{μ} components on the four highlighted orbit legs of RBSP A, plotted as a function of *L*. The power spectra are computed using a 15-min data window, which is moved forward in 2.5 min steps. The orange traces in the bottom two panels labeled $f_{T1_Cummings}$ indicate the model fundamental toroidal wave frequency described in Section 4.2.





Figure 3. Magnetic field observation by RBSP A (red traces) and B (blue traces) at 0830-1230 UT on August 27, 2015. (a) Observatory positions in *L*-MLT coordinates. The thick portions of the RBSP orbits indicate detection of compressional Pc4–5 oscillations. The black arcs indicate the 24 EMMA stations. (b)–(d) Magnetic field components in the local MFA coordinate system based on the T89c model field (Tsyganenko, 1989). Vertical offsets are added to separate data from the two spacecraft. (e, f) *L* and MLAT of the spacecraft. The heavy portion of the line plots indicates the time interval of the compressional Pc4–5 oscillations discussed in the text.





Figure 4. (a)–(e) *L* versus frequency plots of the power spectral density of the full **E**-field and **B**- field components at RBSP B on the outbound leg of orbit 2904. $f_{T1_{LCUMMings}}$ is plotted in brown in each panel. The yellow filled circles in panel (e) indicate the fundamental toroidal wave frequency obtained using the cross-phase analysis of EMMA data. They are plotted as a function of the *L* value of the EMMA stations defined using the T02 model. (f) Electron density.



stations. The yellow filled circles labeled " f_{T1_EMMA} " in Figure 4e indicate the estimated frequency. The *L* values of the EMMA stations are defined using the T02 model (Tsyganenko, 2002). These *L* values are larger than those calculated using the International Geomagnetic Reference Field (IGRF) model (by ~0.1 at $L \simeq 3$ and by ~0.7 at $L \simeq 6$) because of the ring current effect (Berube et al., 2006). We find that f_{T1_EMMA} decreases monotonically from 24 mHz at L = 3.6–11 mHz at L = 5.1 and is well above the center frequency ~7 mHz of the compressional waves. We also examined EMMA data for 1010–1110 UT, which corresponds to the time interval of the symmetric waves at RBSP B (Figure 3d), and found f_{T1_EMMA} at L > 4 was very close to that for 0900–1000 UT. Because EMMA did not fully cover the magnetic field footprint of the spacecraft, we also show $f_{T1_Cummings}$, which is obtained using the n_e data (Figure 4f) as described in Section 3. It is obvious that the compressional waves are not coupled to T1 waves on the local field line. However, we cannot exclude the possibility of coupling of the symmetric waves to P1 waves. The P1 frequency (f_{P1}) is lower than f_{T1} according to cold plasma MHD theory (Cummings et al., 1969), and f_{P1} can be even lower if there is a radial gradient of the ring current thermal pressure (R. E. Denton et al., 2003).

Figure 5 shows the temporal and spectral properties of the compressional waves observed by RBSP B on orbit 2904. The spacecraft was located at L = 4.5 at the center of the time interval shown. The E-field time series indicates that the waves had comparable toroidal (E_{ν}) and poloidal (E_{ϕ}) components with the amplitudes reaching 5 mV/m peak-to-peak (Figure 5a). Magnetic field oscillations appear mostly in the B_{μ} component with a maximum amplitude of 6 nT peak-to-peak (Figure 5b). The E_{ν} - E_{ϕ} hodogram (Figure 5c) shows highly elliptical polarization, with the major axis of polarization tilted ~45° from the ν -axis.

The symmetric waves produce a broad peak at ~6.5 mHz in the PSD of the E_{ν} , E_{ϕ} , and B_{μ} components (Figures 5d and 5e). Additional spectral peaks appear at 12 and 17 mHz, the latter of which coincides with the frequency (the orange downward arrow and vertical dashed line) estimated by interpolating the f_{T1_EMMA} data points shown in Figure 4e to L = 4.5. The frequency 12 mHz at the second spectral peak matches f_{P1} that is obtained by multiplying f_{T1_EMMA} by the theoretical f_{P1}/f_{T1} ratio of ~0.7 (Cummings et al., 1969). We can conclude that the dominant symmetric waves had a frequency much lower than the frequency of the cold plasma fundamental toroidal and poloidal waves excited on the same field line. The $E_{\nu}-E_{\phi}$ coherence is high (Figure 5f) and the $E_{\nu}-E_{\phi}$ cross-phase is ~0 (Figure 5g) in the band occupied by the symmetric waves, meaning that these two components oscillate in phase. The $E_{\phi}-B_{\mu}$ coherence is also high in the wave band (Figure 5h), and the $E_{\phi}-B_{\mu}$ cross-phase is ~-90° (Figure 5i), meaning that the time average of the radial component of the Poynting flux is zero and the wave energy does not propagate radially. This suggests that the mode is radially bound. This is also consistent with the observations that the fluctuations are localized in L.

4.3. Relation to Magnetic Pulsations on the Ground

Figure 6 shows a comparison of the B_{μ} oscillations at RBSP B and magnetic field oscillations detected by EMMA at 1010–1120 UT on August 27. The spacecraft and ground magnetometers were in good conjunction in *L* and MLT during this 70 min interval, as shown in Figure 6a. We can gain information on the azimuthal scale size or the azimuthal wave number (*m*) of the symmetric waves from the space-ground comparison. If the waves have a horizontal scale size at ionosphere height (~100 km) that is comparable to or shorter than the height, they cannot be detected with ground magnetometers (Hughes & Southwood, 1976; Yeoman et al., 2000). This screening effect explains why magnetospheric ULF waves are detected by ground magnetometers only when $|m| \leq 50$ (Takahashi et al., 2013; Wright & Yeoman, 1999; Yamamoto et al., 2018).

In the waveform plots (Figures 6b–6d), we find little similarity between space and ground, and in the spectrum plots (Figures 6e–6g), we do not see any outstanding peaks on the ground in the band occupied by the compressional waves in space (4–9 mHz, shaded yellow). The implication is that the symmetric waves had large (>50) |m| values corresponding to azimuthal wavelength of <4,000 km at the magnetic equator when mapped along the dipole field lines at L = 4.5. However, it is also possible that the waves were highly localized to the equatorial region in the magnetosphere with negligible coupling to the Alfvén mode, in which case there will be no detectable wave fields either at the ionosphere or on the ground regardless of the magnitude of m.

The oscillations on the ground are similar at different locations, so large-scale (|m| < 10) P_{dyn} variations in the solar wind are the most likely driver of the oscillations. At RBSP B, the P_{dyn} -driven oscillations were probably masked by the internally excited symmetric waves.





Figure 5. Time and frequency domain properties of compressional waves detected at RBSP B at 1010–1220 UT on August 27, 2015. The time series is detrended by subtraction of a polynomial as described in the text. The spectra are smoothed by 5-point averaging. (a) Detrended electric field. (b) Detrended magnetic field. (c) Electric field hodogram. (d) Power spectra of the electric field components. The highest peak occurs at 6.5 mHz and is marked by a black dashed line. The orange dashed line marks the frequency (17 mHz) of another spectral peak found in multiple field components, which coincides with f_{T1_EMMA} shown in Figure 4e. The fundamental poloidal wave frequency f_{P1} (= 12 mHz) is estimated by multiplying f_{T1_EMMA} by the theoretical ratio $f_{P1}/f_{T1} \sim 0.7$ (Cummings et al., 1969). (e) Power spectra of the magnetic field components. (f, g) Coherence and cross-phase spectra computed from the E_{ψ} and E_{ϕ} time series. (h, i) Coherence and cross-phase spectra computed from the E_{ϕ} and B_{μ} time series.

5. Possible Wave Excitation Mechanisms

In this section, we examine instabilities that may be relevant to the observed symmetric waves. Source mechanisms external to the magnetosphere are excluded based on the observations indicating that the waves are localized in L (Figure 3) and inferred to have large azimuthal wave numbers (Figure 6). The depression of the background magnetic field during the wave events (Figure 3) also points to internal instabilities associated with enhanced particle pressure.





Figure 6. Comparison of compressional magnetic field oscillations at RBSP B and ground magnetic field oscillations at EMMA. EMMA data for the *L* range (3.7-5.0) of the waves observed by RBSP B are shown in blue. (a) Locations of the spacecraft and EMMA magnetometers for the selected time interval 1010–1120 UT on August 27, 2015. (b)–(d) Magnetic field time series for (b) RBSP B, (c) EMMA station *H* component, (d) and EMMA station *D* component. (e)–(g) Power spectra computed from the detrended time series. The station code and *L* value of the EMMA stations are shown on the right of the spectrum stack plots.



5.1. Drift Resonance Instability

Symmetric poloidal standing Alfvén waves with a finite compressional component can be excited through drift resonance

$$\omega - m\omega_{\rm d} = 0, \tag{1}$$

where ω is the wave frequency and ω_d is the bounce averaged magnetic field gradient and curvature drift frequency of particles. We refer to this mechanism as DRI. We consider ring current ions and westward-propagating waves for this resonance. Theory predicts that the phase of particle flux oscillation changes by ~180° across the resonance energy (Kivelson & Southwood, 1985) and that the resonance transfers energy from ions to poloidal waves in the presence of an inward gradient of the phase space density (\mathcal{F}) of the ions (Chen & Hasegawa, 1991; Southwood, 1976)

$$\left. \frac{\partial \mathcal{F}(L, M, J)}{\partial L} \right|_{M_{\text{res}}, J_{\text{res}}} < 0, \tag{2}$$

where M (magnetic moment) and J are the first and second adiabatic invariants of particle motion with M_{res} and J_{res} being the values for the particles that are in resonance with the waves.

Symmetric poloidal waves and their association with DRI have been reported. Some of the P1 wave studies mentioned in Section 1 (Dai et al., 2013; Takahashi, Claudepierre, et al., 2018; Yamamoto et al., 2018) confirmed the phase shift of ion flux oscillations across the resonance energy and found an inward phase space density gradient. These P1 waves were all observed at geomagnetically quiet times. However, we cannot exclude the possibility that DRI excites symmetric poloidal waves during geomagnetic storms.

We note that particle resonance condition with symmetric waves is generally expressed as

$$\omega - m\omega_{\rm d} = N\omega_{\rm b},\tag{3}$$

where *N* is an even integer and $\omega_{\rm b}$ is the particle bounce frequency (Southwood & Kivelson, 1982). The resonance given by Equation 1 is a special case (*N* = 0) of this condition. We consider the *N* = 0 resonance here with the assumption that the lowest-order resonance contributes most efficiently to the energy exchange between waves and particles. Although the *N* = 0 resonance have been reported for P1 waves as stated above, we are aware that energy exchange can occur through $N \neq 0$ resonances. For a complete evaluation of wave growth, resonance terms with all *N* values need to be considered. Also, even when we consider only the *N* = 0 resonance, a complete instability analysis would include evaluation of the *L* gradient of different $J_{\rm res}$ values corresponding to different equatorial pitch angles. In previous studies, only limited $J_{\rm res}$ values (e.g., $J_{\rm res} \simeq 0$, corresponding to ~90° equatorial pitch angle) have been considered.

5.2. Drift Mirror Instability

The DMI was proposed as a mechanism to generate storm time compressional ULF waves (Lanzerotti et al., 1969). The condition for the instability is

$$\tau \equiv 1 + \sum_{j} \beta_{\perp j} \left(1 - \frac{P_{\perp j}}{P_{\parallel j}} \right) < 0, \tag{4}$$

where *j* indicates ion species, β is the ratio of the thermal to magnetic pressures, *P* is the particle pressure, and the symbols \perp and \parallel indicate directions perpendicular and parallel to the background magnetic field, respectively. The initial DMI theory (Hasegawa, 1969) did not take into account the dipole magnetic field geometry or the effect of wave reflection at the ionosphere. Chen and Hasegawa (1991) updated the theory, including the dipole geomagnetic field, coupling of DMI to field line eigenmodes (standing Alfvén waves), and ion bounce motion, and suggested that the coupled drift Alfvén ballooning mirror (DABM) instability excites antisymmetric waves.

In a recent study using RBSP B data, Soto-Chavez et al. (2019) reported that DMI excited a magnetically compressional ULF wave at ~7 mHz. The wave was observed at L > 5 near the magnetic equator (MLAT $\simeq -0.7^{\circ}$),



in the evening sector, and during the main phase of a moderate geomagnetic storm (Dst minimum = -79 nT). The value of β_{H^+} obtained using RBSPICE data was high (>1). No analysis was presented on the mode structure (symmetric or antisymmetric) along the background magnetic field.

It is possible that the wave reported by Soto-Chavez et al. (2019) had an antisymmetric structure. The wave had a considerable δB_{ν} component unlike our symmetric waves but similar to the antisymmetric stormtime Pc5 waves reported by Takahashi et al. (1987), where the symbol δ indicates a perturbation. But the equatorial node of antisymmetric waves can be shifted away from the dipole equator when the dipole tilt angle is large, especially on the nightside. The tilt angle during the Soto-Chavez et al. (2019) event was large (~30°), and we suspect that the equatorial node of the wave was not located exactly at the dipole equator (MLAT = 0). In fact, the TS05 magnetic field model (Tsyganenko & Sitnov, 2005) places both the minimum magnetic field magnitude and the field line maximum distance at MLAT = -2.2° for the field line that passes the spacecraft at the epoch of the maximum wave activity (1405 UT on July 6, 2014). If the wave symmetry can be defined with respect to this modified equator, the spacecraft was located 1.5° north of the node. In this case, simultaneous detection of δB_{μ} and δB_{ν} of an antisymmetric wave is possible according to Figures 3 and 5 of Takahashi et al. (1987), and the DABM instability is a viable excitation mechanism.

5.3. Drift Compressional Instability

Compressional magnetospheric waves are excited by an instability when the radial gradients of plasma bulk parameters satisfy certain conditions. We refer to this instability as the DCI. DCI theory was initially developed using a slab geometry for the plasma and magnetic field (Hasegawa, 1971a, 1971b). Rosenbluth (1981) discussed the instability mechanism incorporating the effects of trapped particles. Ng et al. (1984) considered trapped particles and bounce averaging in a dipole magnetic field and used a symmetric trial function to study the instability. Cheng and Lin (1987) studied both DMI and DCI without setting the field line mode structure a priori and showed that the former has an antisymmetric structure whereas the latter has a symmetric structure. Crabtree et al. (2003) studied DCI in realistic magnetic fields in the drift-kinetic approximation and showed that there are two types of instabilities, which are described below. Crabtree and Chen (2004) added finite Larmor radius effects and performed a rigorous analysis of how the compressional modes decoupled from the electrostatic and shear Alfvén modes. P. N. Mager et al. (2013) studied the effect of a bump-on-tail ion energy distribution function on DCI.

Some observational studies considered DCI as a possible source mechanism for ULF waves. C. A. Green (1985) suggested DCI for plasmaspheric giant pulsations observed on the ground. More recently, the instability was considered for symmetric poloidal waves observed by the Cluster spacecraft in the dayside magnetosphere at L = 4-6 during a geomagnetically quiet period (Eriksson et al., 2005), compressional Pc5 waves observed by the five Time History of Events and Macroscale Interactions during Substorms Mission spacecraft at $L \simeq 10$ in the dusk sector (Rubtsov et al., 2018), and nighttime Pc5 waves detected by radar (Chelpanov et al., 2016; Chelpanov et al., 2018; O. V. Mager et al., 2019). One of the radar events, which was detected at L = 4.6-7.8 in the postmid-night sector at ~1.8 mHz and exhibited $m \simeq -10$, was also detected by RBSP (O. V. Mager et al., 2019). It is questionable that this radar event was of the same origin as our symmetric waves because our waves had higher frequencies and likely much larger |m|.

We provide a brief review of key DCI properties obtained by Crabtree et al. (2003) and Crabtree and Chen (2004) that are relevant to our spacecraft observations. Although their studies were motivated by Pi2 waves excited on the nightside, the results can be applied to dayside ULF waves as well. The theory assumed hot ions (\sim 10 keV protons) to be the energy source.

DCI occurs in two situations. The first is when the pressure gradients become sufficiently steep to reverse the magnetic-guiding center drift (referred to as DCI condition 1). In the dayside inner magnetosphere where the magnetic field is dominated by the dipole component, the ion magnetic field gradient and curvature drift are westward. Therefore, for ions to satisfy DCI condition 1 (westward diamagnetic drift), the gradient of the ion pressure (P) needs to be outward,

$$\partial P/\partial L > 0.$$

(5)



The second situation is when the ion temperature (T) gradient is in the opposite direction to the ion density (n) gradient (referred to as DCI condition 2). This condition can be expressed as

$$\eta \equiv \mathcal{L}_n / \mathcal{L}_T < 0, \tag{6}$$

where $\mathcal{L}_n (\equiv (\partial \log n/\partial L)^{-1})$ and $\mathcal{L}_T (\equiv (\partial \log T/\partial L)^{-1})$ are the density and temperature gradient scale lengths with the sign included. For fixed η , there is a range of $\beta_{low} < \beta < \beta_{high}$ that is unstable, and for a fixed β , there is a range of $\eta_1 < \eta < \eta_2 < 0$ that is unstable. For $\eta_i < -0.5$, both β_{low} and β_{high} become smaller for more negative η_i (see Figure 1 of Crabtree and Chen (2004) for the region in parameter space for instability).

The basic understanding of the physical mechanism of instability for the DCI was first given by Rosenbluth (1981) and is similar to the physics that drives the mirror modes (Southwood & Kivelson, 1993). Here, we give a short analogy of DCI to DMI. At low frequencies, the perturbed particle pressure δP_{\perp} is a function of the magnitude of the magnetic field perturbation δB_{μ} . In a homogeneous anisotropic plasma, relevant to DMI, the perturbed pressure for the bulk of the particles is given as

$$\delta P_{\perp \text{bulk}} \simeq 2P_{\perp} \left(1 - T_{\perp}/T_{\parallel}\right) \delta B_{\mu}/B,\tag{7}$$

where B is the magnitude of the magnetic field (Southwood & Kivelson, 1993).

However, resonant particles, with zero parallel velocity for purely growing modes, are different, and they respond in such a way that the perturbed pressure is in phase with the magnetic perturbation for growing waves. In fact, it is possible to express the resonant particle contribution to the perturbed pressure as

$$\delta P_{\perp \rm res} \simeq 2 \frac{\gamma}{k_{\parallel}} n_{\rm res} \frac{T_{\perp}^2}{T_{\parallel}} \frac{\delta B_{\mu}}{B},\tag{8}$$

where γ is the growth rate, k_{\parallel} is the wavenumber parallel to the magnetic field, and n_{res} is the density of resonant particles. Then the change in energy due to a perturbation near marginal stability is

$$\delta W_{\rm DMI} \simeq \left\{ \frac{B^2}{\mu_0} + 2P_{\perp} \left(1 - \frac{T_{\perp}}{T_{\parallel}} \right) + 2\frac{\gamma}{k_{\parallel}} n_{\rm res} \frac{T_{\perp}^2}{T_{\parallel}} \right\} \frac{\delta B_{\mu}}{B} \simeq 0.$$
(9)

To maintain pressure balance when $T_{\perp}/T_{\parallel} > 1$, where the bulk of the particles are responding out of phase with the magnetic field, the perturbed magnetic field must be growing ($\gamma > 0$).

DCI is similar except that the perturbed pressure of the nonresonant particles is now given as

$$\delta P_{\perp \text{bulk}} \simeq (\partial P / \partial L) (\partial B / \partial L)^{-1} \delta B_{\mu}. \tag{10}$$

Then if the product of the two L derivatives on the right-hand side of Equation 10 is negative, which means that the magnetic and pressure perturbations are out of phase, we see that the pressure gradients and magnetic gradients in physical space play the same role as the temperature anisotropy in mirror modes. The resonant particles in the DCI are resonant with the bounce-averaged magnetic guiding center drift of the ions (rather than the parallel motion), and there is a real frequency component to the DCI. Thus, we find an equation similar to the drift mirror modes,

$$\delta W_{\rm DCI} \simeq \left\{ \frac{B^2}{\mu_0} + (\partial P/\partial L)(\partial \log B/\partial L)^{-1} + P_{\rm \perp res} \right\} \frac{\delta B_{\mu}}{B} \simeq 0.$$
(11)

In Crabtree and Chen (2004), the resonant particle contribution is found by solving the gyrokinetic equation, but physically the role of the resonant particles is the same. They respond in phase to the magnetic perturbation only in growing waves, and the balancing of the total pressure leads to the instability.

Because the frequencies are low, one must consider the bounce-averaged motion along the field lines, as was done in Crabtree and Chen (2004) and Cheng and Lin (1987). For δB_{μ} eigenmode structures that are antisymmetric around the magnetic field, the resonant particle contribution to the perturbed pressure vanishes by bounce averaging. Because DCI is a resonant particle instability, as explained above, DCI should have a symmetric eigenmode structure.



6. Examination of Instability Conditions

We examine whether DRI, DMI, or DCI is relevant to the symmetric waves using proton measurements made at RBSP A on orbit 2920 with the HOPE and RBSPICE instruments. These instruments detected He⁺ and O⁺ also, but we do not include these ions in our analysis because their contribution to the plasma pressure was much smaller than H⁺. To indicate that we are using only H⁺ data, we add the subscript "H⁺" to ion plasma parameters.

6.1. Proton Data Overview

Figure 7 provides an overview of the proton data. The orbit occurred during a period of moderate auroral electrojet activity (Figure 7a), so injection of particles in the midnight sector is expected. The frequency-time spectrogram of the B_{μ} component (Figure 7b) indicates symmetric waves on the outbound leg at 0900–1030 UT (L = 3.8-5.3). Compressional waves are detected also at large distances, from 1110 UT (L = 5.6) to 1320 UT (L = 5.4) encompassing the apogee (1200 UT, L = 5.8), with broader spectral bandwidths than the symmetric waves. We do not discuss the mode structure or the generation mechanism of these large-L waves. No notable waves are seen at L < 5.3 on the inbound leg. The difference between the two orbital legs can be either temporal or spatial. Because the spacecraft crossed L = 4.5 at 14.1 hr MLT on the outbound leg (waves detected) and at 18.2 hr MLT on the inbound leg (no waves detected), it is possible that the waves were limited to MLTs earlier than 18 hr.

Figure 7c is an energy (W)-time spectrogram of the proton differential energy flux (j_{H^+}) at 90° pitch angle, generated by combining data included in the HOPE and RBSPICE level-3 data products (ect-hope-PA-L3 and rbspice_lev-3-pap_tofxeh). The protons appearing in this figure form three populations, referred to as "low-energy plasmasphere," "energized in situ population," and "injected high-energy plasma sheet," following M. H. Denton et al. (2016). The injected population is explained by drift of plasmasheet ions that are injected from the nightside into drift orbits passing the L < 6 region, but the in situ population cannot be explained by this mechanism. Possible mechanisms to produce the energized in situ population include wave-particle interaction involving ion-acoustic waves, Alfvén waves, or ion cyclotron waves (M. H. Denton et al., 2016). The symmetric waves were detected outside the plasmasphere, so the latter two populations are of interest to us. The two populations are separated by a demarcation energy of \sim 5 keV. The energized in situ population starts from the lower energy limit of the instrument (~ 0.001 keV) and extends to a cutoff, which changes from ~ 3 keV at 0900 UT to < 0.1 keV after 1100 UT. This population exhibits high flux values and high cutoff energies when the symmetric waves are present, but it is much weaker on the outbound leg, which may explain the absence of symmetric waves. The high-energy plasma sheet population forms narrow horizontal structures above ~ 10 keV. Within this population, there is a time delay in the appearance of lower-energy protons (energy dispersion), which is attributed to the energy dependence of the magnetic gradient and curvature drift speed of particles injected on the nightside. The high-energy plasma sheet population is present on both the outbound and inbound legs with similar energies and intensities.

Figure 7d compares j_{H^+} versus energy plots for the two 20 min intervals marked in Figure 7b. At $W < \sim 5$ keV, j_{H^+} is significantly higher in the first interval (0920–0940 UT, red curve) than in the second interval (1420–1440 UT, blue curve). At $W > \sim 5$ keV, the difference is small. This suggests that protons at $W < \sim 5$ keV play an important role in exciting the symmetric waves.

Figure 7e shows phase space density plots for the same time intervals. An important feature to note is the slope $\frac{\partial F_{H^+}}{\partial W}$ at W > 1 keV. The high-energy population that forms a peak in the j_{H^+} plots produces a plateau (outbound leg) or a mild positive slope (inbound leg) for \mathcal{F}_{H^+} (Figure 7e) in the energy range 3–20 keV. A positive slope (or a bump on tail phase space density) is a possible free energy for bounce resonance excitation of P2 waves (Hughes et al., 1978). The absence of a positive energy slope on the outbound leg implies that P2 waves were not destabilized, which is consistent with the dominance of symmetric waves.

6.2. Drift Resonance Instability

Assuming that the symmetric waves propagate westward and can satisfy the resonance condition given by Equation 1 at some energy, we examine whether an earthward phase space density gradient (Equation 2) was present to make symmetric waves unstable.





Figure 7. (a) Auroral electroject indices for August 27, 2015. (b) Dynamic spectra of the B_{μ} component. The red horizontal bar at the bottom indicates the 0920–0940 UT interval (L = 4.2–4.6) on the outbound leg when the intensity of symmetric waves was high. The blue horizontal bar indicates the time interval 1420–1440 UT on the inbound leg when the spacecraft moved from L = 4.6 to L = 4.2 but did not detect symmetric waves. These time markers are also shown in panel (a). (c) Energy-time display of proton differential energy flux at 90° pitch angle measured by HOPE and RBSPICE. (d) Proton differential energy flux averaged over the time intervals shown by the red and blue time markers The vertical line indicates the energy boundary (~53 keV) between HOPE and RBSPICE. The shading indicates energies (<30 eV) that are not included in calculating the proton bulk parameters shown in Figure 9. (e) Same as (d) but for the phase space density.

Figure 8 shows properties of protons that are relevant to DRI. Assuming that equatorially mirroring particles interact with the symmetric waves most effectively, we evaluate \mathcal{F}_{H^+} at $J_{res} = 0$. On the selected RBSP A orbit, the equatorially mirroring particles are nearly equivalent to particles having local pitch angles of ~90° because the spacecraft was very close to the magnetic equator ($|MLAT| < 3.5^\circ$) at L > 3. Therefore, we evaluate \mathcal{F}_{H^+} at 90° pitch angle at the spacecraft. As for M_{res} , we consider six trial values, 0.01, 0.02, 0.05, 0.1, 0.2, and 0.5 (keV/ nT). At L = 4.5, where the symmetric wave was strong, the equatorial dipole field magnitude 330 nT gives the corresponding proton energies of 3.3, 6.6, 16.5, 33, 66, and 165 keV. For these energies, we use the approximate





Figure 8. Proton parameters relevant to DRI, derived from measurements made at RBSP A on orbit 2920. (a) Proton energies corresponding to the six trial values of magnetic moment $M_{\rm res}$ defined using the magnitude of the measured magnetic field. The yellow shading indicates the time interval of symmetric waves. (b) Variation of proton phase space density ($\mathcal{F}_{\rm H^+}$) evaluated at the selected magnetic moments. We define $\mathcal{F}_{\rm H^+}$ using 10 min averages of $j_{\rm H^+}$. (c) The same $\mathcal{F}_{\rm H^+}$ values plotted as a function of dipole *L* for the outbound and inbound legs.

formula by Hamlin et al. (1961) to find the guiding center drift frequency (ω_{dH^+}) and then obtain the azimuthal wave number (m, negative for westward propagation) that satisfies the resonance condition (Equation 1) by setting the wave frequency at 7 mHz to represent the observation (see Figure 5). The m values, shown in Figure 8, vary from -1,200 for $M_{res} = 0.01$ keV/nT to -25 for $M_{res} = 0.5$ keV/nT. The |m| value of 1,200 is much larger than those reported for Pc5 waves (Takahashi et al., 1985; Wright & Yeoman, 1999). However, this |m| translates to an azimuthal wavelength of 150 km, which is still larger than the gyroradius (~30 km) of the resonant (3.3 keV) protons.

Figure 8a shows time series of the proton energies corresponding to the trial M_{res} values plotted in 10 min steps. To get the energies (= $M_{\text{res}}B$), we used the measured *B*, which is slightly (~10%) lower than the dipole value. The yellow shading indicates where the symmetric waves were detected. We have marked the energy limits of HOPE and RBSPICE to show that these instruments cover the six energies except near perigee. The corresponding $\mathcal{F}_{\text{H}^+}(M_{\text{res}}, J_{\text{res}})$ time series, also given in 10 min steps, is shown in Figure 8b. There is a sharp decline in \mathcal{F}_{H^+} as the spacecraft approaches the perigee. Elsewhere, the behavior of \mathcal{F}_{H^+} depends on M_{res} .

The same \mathcal{F}_{H^+} data are plotted as a function of dipole *L* for the outbound leg (Figure 8c) and the inbound leg (Figure 8d). Figure 8c indicates an inward gradient of \mathcal{F}_{H^+} occurring in the *L* domain of the symmetric waves for the three lowest M_{res} values corresponding to the resonance energy (at L = 4.5) of ≤ 16.5 keV and $|m| \geq 250$. This



result indicates that DRI is a possible mechanism for the symmetric waves. The \mathcal{F}_{H^+} profile for the inbound leg is not as clear, but there is an indication of an inward gradient for the three lowest M_{res} values. Despite the inward gradient, no symmetric waves were detected on the inbound leg. Perhaps the gradient was not steep enough or the magnitude of \mathcal{F}_{H^+} was not high enough to excite symmetric waves.

6.3. Drift Mirror Instability

Figure 9 contains proton bulk parameters relevant to DMI along with the B_{μ} dynamics spectra, the magnetic field magnitude, and the electron density (top three panels) and the spacecraft position (bottom three panels). The number density ($n_{\rm H^+}$, Figure 9d) and pressure ($P_{\rm H^+}$, Figure 9f) are obtained from the moments of $j_{\rm H^+}$ measured by HOPE and RBSPICE. The proton temperature ($T_{\rm H^+}$, Figure 9e) is given by $T_{\rm H^+} = P_{\rm H^+}/(k_{\rm B}n_{\rm H^+})$, where $k_{\rm B}$ is the Boltzmann constant. The moment calculation uses HOPE data in the energy range 30 eV to 51.8 keV and RBSPICE data in the energy range 54.7–598 keV. Although HOPE detects ions with energies as low as 1 eV, the moment calculation uses data for >30 eV to avoid possible errors arising from spacecraft charging.

It is clear that DMI is an unlikely mechanism for the symmetric waves. Figure 9k shows the instability parameter τ_{H^+} defined by Equation 4. Although the temperature anisotropy $T_{\perp H^+} > T_{\parallel H^+}$ (Figure 9e) is qualitatively favorable for the instability, τ remains >0 throughout the orbit because β_{H^+} (Figure 9j) is not high enough to bring τ down to the instability regime (<0, shaded gray in Figure 9k).

6.4. Drift Compressional Instability

DCI is a more promising mechanism for the symmetric waves. Included in Figure 9 are the parameters relevant to DCI defined in Section 5.3: $\mathcal{L}_{n_{\rm H^+}}$, $\mathcal{L}_{T_{\rm H^+}}$, and $\eta_{\rm H^+}$. We calculated $\mathcal{L}_{n_{\rm H^+}}$ and $\mathcal{L}_{T_{\rm H^+}}$ after smoothing the $n_{\rm H^+}$ and $T_{\rm H^+}$ time series by 55-point (20 min) running average. Two time intervals of sustained negative $\eta_{\rm H^+}$ are evident, and we highlight them by shading in yellow (outbound leg) and blue (inbound leg). In the time interval shaded yellow, $n_{\rm H^+}$ decreased with L (Figure 9d), while $T_{\rm H^+}$ increased (Figure 9e). The median values during the symmetric wave event are $\mathcal{L}_{n_{\rm H^+}} \sim -1.3 R_{\rm E}$, $\mathcal{L}_{T_{\rm H^+}} \sim 1.7 R_{\rm E}$. As a result, we get $\eta_{\rm H^+} \sim -0.6$, which satisfies DCI condition 2 (Equation 6).

We can make a quantitative evaluation of the frequency of waves generated by DCI. First, the theory by Crabtree and Chen (2004) predicts that the wave frequency (ω) is on the order of the proton diamagnetic drift frequency (ω_{*H^+}), expressed as

$$\omega_{*\mathrm{H}^{+}} = k_{\phi} \rho_{\mathrm{H}^{+}} v_{\mathrm{H}^{+}} / \mathcal{L}_{n_{\mathrm{H}^{+}}}, \qquad (12)$$

where k_{ϕ} is the angular azimuthal wave number, ρ_{H^+} is the proton Larmor radius, and v_{H^+} is the proton velocity. Then including the temperature gradient into the diamagnetic drift frequency and assuming the distribution is approximately Maxwellian (locally), we get

$$\omega \simeq \omega_{*H^+} (1 + 1.5\eta_{H^+}). \tag{13}$$

Using the $T_{\perp H^+}$ value of ~15 keV (Figure 9e), we obtain $\rho_{H^+} \simeq 50$ km and $v_{H^+} \simeq 1,700$ km/s, and we get $|\mathcal{L}_{n_{H^+}}| \simeq 1.3 R_E$ from Figure 9g. Because the remaining parameter k_{ϕ} on the right-hand side of Equation 12 cannot be determined observationally, let us first assume $|m| \simeq 100$, which is a reasonable value according to the discussion of the ionospheric screening effect presented in Section 4.3. Then for $L \simeq 4.5$ (an approximate center location of the symmetric waves), we get $k_{\phi} \simeq 100/4.5 \simeq 22$ rad R_E^{-1} . From these, we get $k_{\phi}\rho_{H^+} \simeq 0.2$, a reasonable value according to Crabtree and Chen (2004), and from Equations 12 and 13, we get $\omega \simeq 0.005$ rad Hz or ~1 mHz.

The ω value obtained above is about an order of magnitude too small for our observations and definitely lower than the local P1 wave frequency. But if $\beta_{H^+} \ll \omega_{dH^+}/\omega_{*H^+}$, then the frequency is closer to the bounce averaged drift frequency of the ions. A simplified explanation for this is that if β_{H^+} is small enough, then the resonant contribution has to be closer to the thermal core of the ions to get the resonant contribution in order to drive the instability. This makes the mode behave more like an "energetic particle mode" or beam-like mode. If β_{H^+} is large enough, then a small fraction of resonant particles is sufficient to drive the instability and the wave takes on the





Figure 9.



characteristic of the neutrally stable drift mode. So if $\beta_{\rm H^+} \simeq 0.5$, then $\omega_{\rm dH^+}$ is about an order of magnitude higher to bring ω closer to the observed ~7 mHz. Another way to bring theoretical ω closer to the observation is to assume a much larger |m| value, for example, $|m| \simeq 500$, in which case we have $k_{\phi}\rho_{\rm H^+} \simeq 1$. However, such a high |m| number has not been observed, as we pointed out in Section 6.2.

DCI condition 2 ($\eta_{H^+} < 0$) also occurred on the inbound leg of the same orbit at 1420–1520 UT (shaded blue), when the spacecraft was close to the magnetic equator ($|MLAT| < 3^\circ$, Figure 9m) and in an *L* range (2.9–4.4, Figure 9l) similar to that during the symmetric wave event on the outbound leg. However, no notable B_{μ} oscillation was detected on the inbound leg. The reason for this difference might be that the n_{H^+} and T_{H^+} gradients each had a sign opposite that during the first interval, the high electron density, or the low intensity of the thermal population.

In relation to the \mathcal{F}_{H^+} plot shown in Figure 7e, we point out that the theoretical DCI analysis (Crabtree & Chen, 2004; Crabtree et al., 2003) assumed a single Maxwellian for the ions. The shape of the observed \mathcal{F}_{H^+} is not a single Maxwellian, which implies that there are multiple gradient scale lengths, which would complicate stability analysis. However, we believe that the instability occurs even when more than one gradient scale exists. We argue that the theory gives a conservative estimate of the instability threshold when we use the bulk parameters derived for ions having a structured phase space density because a structured phase space would only add to the free energy available to drive the instability.

6.5. Caveats

Our analysis of DRI and DCI is incomplete in two aspects. First, there is ambiguity between spatial and temporal variations in the structure of the background plasma. What we defined as a radial gradient or a radial gradient scale length could have significant error due to temporal effects such as energy dispersion of ion injections or changes in solar wind dynamic pressure. For the August 27 event examined in detail, the RBSPICE data were missing from RBSP B, which makes the RBSP A and B comparison difficult.

Second, we lack some observations that are crucial to unambiguously determining the wave generation mechanism. One is the azimuthal wave number (including both the magnitude and the sign). This means that we cannot determine $k_{\phi}\rho_{H^+}$, an important parameter in the DCI theory. The *m* value also determines whether drift resonance (Equation 1) is possible. Another is the cross-phase between δB_{μ} and δj_{H^+} , which provides a test of drift resonance. The cross-phase will change by ~180° across the resonance energy in association with a peak amplitude of δj_{H^+} (Dai et al., 2015). Although RBSP carried instruments to detect ions at relevant energies, they were not sensitive enough to determine δj_{H^+} during the selected symmetric wave events.

7. Electron Response

This section examines the relationship between symmetric waves and energetic electrons. ULF waves with both low-*m* and high-*m* numbers can interact with energetic electrons. A general theoretical description has been given by Kivelson and Southwood (1985). Mann et al. (2013) presented evidence that low-*m* (~2) waves can interact with energetic electrons through drift resonance. Claudepierre et al. (2013) reported electron resonance with a P1 wave having an intermediate *m* value of ~40. Ukhorskiy et al. (2009) proposed that high-*m* waves with a broad-*m* spectrum can lead to substantial electron energization and radial transport. Zong et al. (2017) presented a comprehensive review on the subject.

Figure 9. Relationship between compressional magnetic field oscillations and quantities related to DMI and DCI, shown for RBSP A orbit 2920. (a) Dynamic spectra of the B_{μ} component. (b) Magnetic field magnitude. (c) Electron number density. (d)–(k) Parameters derived from proton fluxes measured by the HOPE and RBSPICE experiments. See text for definition. Some parameters are color coded to indicate the components perpendicular (red) and parallel (blue) to the background magnetic field. The areas shaded yellow or blue indicate sustained intervals of negative η_{H^+} . The gray shading indicates the parameter domain satisfying the condition for DCI or DMI. (l–n) Magnetic coordinates of the spacecraft.





Figure 10. Overview of the relationship between electron phase space density and symmetric waves. (a) Magnetopause standoff distance according to Shue et al. (1998). The solar wind OMNI data are used as input. (b) Electron phase space density at RBSP A evaluated for $M_e = 109$ MeV/G and $J_e = 0.11 R_E G^{1/2}$ and plotted as a function of time and L^* . Each pixel covers a half orbit (4.5 hr) along the time axis. The rectangles drawn with black dotted lines indicate the symmetric compressional waves identified in Figure 2. (c) Dst index.

7.1. Electron Flux Phase Space Density

To understand whether the symmetric waves play any role in the transport of electrons in the outer radiation belt, we first examine the spatial and temporal relationship between the waves and electrons. Figure 10 shows the evolution of the phase space density (\mathcal{F}_e) of energetic electrons at RBSP A (Figure 10b) along with a model magnetopause standoff distance (Figure 10a) and Dst (Figure 10c) during the selected storm interval. \mathcal{F}_e is obtained using the standard definition $\mathcal{F}_e = j_e/p_e^2$, where p_e is the electron momentum (Hilmer et al., 2000). The illustrated \mathcal{F}_e is evaluated at M = 109 MeV/G and $J = 0.11 R_E G^{1/2}$ and plotted as a function of time and L^* , where L^* the third adiabatic (magnetic flux) invariant.

The range of L^* covered by RBSP A during the detection of symmetric waves is marked by a rectangle drawn for orbits 2920, 2922, and 2925 (see Figure 2). Here we emphasize that compressional ULF waves were detected by RBSP A on every orbit from 2917 (August 26) to 2925 (August 29), as shown in Figure 2g. It is likely that the waves had a symmetric structure on all or some of these orbits. We find that the inner L^* edge of the symmetric waves is nearly colocated with the L^* region where \mathcal{F}_e exhibits a steep outward gradient. In this region, the waves may diffuse the electron inward by violating the L^* values of the electrons and move the inner edge of the radiation belt closer to Earth (Green & Kivelson, 2001).

7.2. Electron Flux Oscillations

Although it is beyond the scope of the present study to determine the role that the symmetric waves might play in redistributing radiation belt electrons over the entire L shells covered by the spacecraft, we examine the response of electrons to the waves using observation made at $L \sim 5$. An important target of this analysis is drift





Figure 11. Electron flux variations during the symmetric wave event at RBSP A on orbit 2920. (a) Electric field. (b) Magnetic field compressional component. (c) Electron flux at 90° pitch angle, smoothed by averaging over three spins (33 s). Vertical dashed lines are drawn through the peaks, most clearly seen at 33 and 54 keV. (d) Energy dependence of the electron phase space density at six pitch angles averaged over 1010–1020 UT. (e) Pitch angle dependence of the electron phase space density shown at multiple energies.

resonance (Equation 1), which is the elementary process of energy exchange between ULF waves and electrons in the context of radiation belt dynamics (Elkington et al., 1999; Fälthammar, 1965).

Two conditions are necessary for the drift resonance to occur. First, the waves need to propagate eastward (m > 0), the same direction as the electron guiding center drift. Second, |m| needs to have an appropriate value to satisfy the resonance condition. For example, |m| must be in the range 4–120 for the electrons in the MagEIS energy range considered here (33 keV–1.6 MeV) to experience the resonance with a 7 mHz wave excited at L = 4.5, when the particle drift speeds are evaluated using the dipole magnetic field (Hamlin et al., 1961).

We examine the temporal variation of the electron differential energy flux (denoted j_e) during the wave event to gain insight into the resonance. Figure 11 shows the relationship between the symmetric waves and j_e observed at RBSP A on orbit 2920. Figures 11a and 11b show the E_{ν} , E_{ϕ} , and B_{μ} components of the waves, and Figures 11c shows j_e at 90° pitch angle. Oscillations are visible at all energies but most clearly at the two lowest energies



(the irregular variations at high energies are attributed to low count rates). By following the vertical dashed lines drawn through the peaks of j_e at 33 keV, we find that j_e oscillates in phase with B_{μ} and ~90° out of phase with E_{ν} and E_{ϕ} .

Figure 11 includes information on the dependence of the equilibrium \mathcal{F}_e on *L*, pitch angle, and kinetic energy, which is necessary to determine the cause of particle flux oscillations in ULF wave fields (Kivelson & Southwood, 1985; Southwood & Kivelson, 1981). Figures 11c indicates that j_e decreased as the spacecraft moved outward. This is interpreted to be a spatial effect, that is, j_e (and \mathcal{F}_e) had an inward gradient. Figures 11d shows that \mathcal{F}_e rapidly decreases as energy increases. Figures 11e shows that \mathcal{F}_e is peaked at 90° pitch angle.

Figure 12 shows the detail of the j_e oscillations and their relation to the symmetric waves. We represent j_e oscillations by the detrended version of its logarithm, denoted log j_e . Figures 12a–12c indicate that log j_e has a phase difference of ~90° from E_{ϕ} and ~0 from B_{μ} , respectively, regardless of energy or pitch angle. This is confirmed in Figures 12g and 12i, which show the E_{ϕ} and log j_e spectral parameters evaluated at the wave frequency 5.5 mHz (Figure 12e). The E_{ϕ^-} log j_e cross-phase is ~90° regardless of energy (Figure 12g) or pitch angle (Figure 12j).

There are a few points to be noted from Figure 12. First, we do not see evidence for electron drift resonance. At the resonance energy, $\log j_e$ PSD would be peaked and the E_{ϕ} - $\log j_e$ and B_{μ} - $\log j_e$ cross-phase would exhibit a 180° shift (Claudepierre et al., 2013; Mann et al., 2013). Obviously, the waves were propagating westward (m > 0), opposite to the direction of the electron magnetic field gradient and curvature drift or, for the case of m > 0, the resonance energy was outside the energy range of the electron data examined, that is, m < 4 or m > 120.

To understand the cause of the j_e oscillations, we start from a general expression for the perturbation of particle phase space density

$$\delta \mathcal{F} = -M \delta B_{\mu} \frac{\partial \mathcal{F}}{\partial M} - \delta W \frac{\partial \mathcal{F}}{\partial W} - \delta L \frac{\partial \mathcal{F}}{\partial L},\tag{14}$$

where \mathcal{F} is a function of L, M, and W and δW and δL are the perturbations of W and L experienced by individual particles (Southwood, 1973). Both δW and δL depend on the spatial and temporal structure of the waves and on the energy and pitch angle of the particles. Therefore, δW and δL include effects of drift-bounce resonances and need to be evaluated by taking an integral over the trajectory of the particle guiding center. Based on the absence of resonance signatures, we pay attention only to particle modulation mechanisms that do not include resonance effects.

The first effect is radial convection of the gradient of the phase space density. The phase space density perturbation arising from the convection, denoted δF_c , is part of the δL term in Equation 14 (Southwood, 1973) and is given as

$$\delta \mathcal{F}_{\rm c} = -\xi_{\rm v} \frac{\partial \mathcal{F}}{\partial L},\tag{15}$$

where ξ_{ν} is the radial component of field line displacement. This represents convection of the phase space density by the radial motion of the flux tube induced by E_{ϕ} . Using the electron data shown in Figure 11, we obtained $\mathcal{F}_{e}(M, W)$ at different time (L) steps and confirmed $\frac{\partial F_{e}}{\partial L} < 0$. Here the independent variables are converted from W and pitch angle to W and M using the magnitude of the measured magnetic field. According to Figures 13a and 13b, ξ_{ν} associated with a symmetric wave reaches a maximum when the E_{ϕ} perturbation crosses zero from positive to negative. At the zero crossing, δF_{e} (and the j_{e} perturbation) observed by a spacecraft reaches a maximum if \mathcal{F}_{e} has an inward gradient. This predicted E_{ϕ} - log j_{e} phase relationship is just the opposite of what is observed, and we exclude the convection of gradient as the cause of the observed j_{e} oscillations.

Another non-resonance mechanism arises from the temporal variation of the magnetic field seen from drifting particles and is related to the M and W dependence of \mathcal{F} . Kivelson and Southwood (1985) called this mechanism the betaton effect and expressed it as

$$\delta \mathcal{F}_{\rm b} = -M \delta B_{\mu} \left(\frac{1}{B} \frac{\partial \mathcal{F}}{\partial M} + \frac{\partial \mathcal{F}}{\partial W} \right). \tag{16}$$



Figure 12.





Figure 13. (a) Radial field line displacement pattern of a symmetric wave plotted a quarter-wave period apart for one wave period (adopted from Figure 2a of Takahashi et al. [2011]). The red arrows indicate the plasma bulk velocity. (b) Schematic time series plots of the perturbations of observed field components (E_{ϕ} and B_{μ}) and the electron flux (j_e) during the symmetric wave event shown in Figure 12 along with the inferred field line displacement (ξ_{ν}) and the electron flux perturbation associated with it ($j_{e, \text{ conv}}$). The phase of $j_{e, \text{ conv}}$ reflects the illustrated ξ_{ν} and the observed inward gradient of \mathcal{F}_e . (c) \mathcal{F}_e derived from the j_e measurements made by RBSP A MagEIS at 1010–1020 UT, plotted as a function of *M* and *W*. The black dots indicate the data points corresponding to the energies and pitch angles of the measurements.

We can determine the relative importance of the two terms on the right-hand side of Equation 16 using the contour display of \mathcal{F}_e in the *M*-*W* space, shown in Figure 13c. The \mathcal{F}_e values for this figure are defined using the j_e data averaged over 1010–1020 UT (Figures 11d and 11e). We find $\frac{\partial \mathcal{F}_e}{\partial M} > 0$ and $\frac{\partial \mathcal{F}_e}{\partial W} < 0$ in the *M*-*W* space covered by the MagEIS data. Because only the second derivative leads to $\delta \mathcal{F}_e > 0$ (in-phase oscillations of B_{μ} and j_e), we conclude that the second term on the right-hand side of Equation 16 makes the dominant contribution to the j_e oscillations.

It is interesting to note that previous studies of compressional Pc5 waves at geostationary satellites reported antiphase B_{μ} - j_e oscillations when j_e had a pancake pitch angle distribution (Kremser et al., 1981; Takahashi et al., 1985). The first term on the right-hand side of Equation 16 probably played the major role in modulating j_e during these Pc5 wave events.

To conclude the electron data analysis, we find no evidence of drift resonance with the symmetric waves. However, we caution the readers that we conducted the analysis of j_e oscillations for a time interval when the spacecraft was close to the outer L limit of the symmetric wave activity and detected an inward gradient of \mathcal{F}_e . MagEIS data for lower L do not show clear j_e oscillations, and the electron response in that crucial region remains to be understood.

Figure 12. Relationship between symmetric waves and electron flux oscillations observed by RBSP A on orbit 2920. (a) Detrended E_{ϕ} time series. (b) Detrended B_{μ} time series. (c) Detrended \log_j_e time series at 13 energies. The pitch angle is fixed at 90°. The vertical dashed lines mark the peaks seen at 33 keV. (d) Detrended \log_j_e time series at 10 pitch angles. The energy is fixed at 33 keV. (e) Power spectral density (PSD) of the data shown in panels (a) and (b). The vertical dashed line marks the spectral peaks at 5.5 mHz. (f) PSD of the data shown in panel (c), evaluated at 5.5 mHz and plotted as a function of energy. (g) Cross-phase between E_{ϕ} and \log_j_e (red) and between B_{μ} and \log_j_e (blue). (h) PSD of the data shown in panel (d), plotted as a function of pitch angle. (i) Cross-phase between E_{ϕ} and \log_j_e (red) and between B_{μ} and \log_j_e (blue).





Figure 14. Illustration of stormtime ULF waves and their relation to ion populations incorporating results from previous and present studies. Antisymmetric waves are excited at L > 5 by energetic ions (high-energy plasma sheet) drifting westward from the nightside injection region. Symmetric waves are excited at L < 5 in the region populated by lower-energy ions (energized in situ population) in addition to the high-energy plasma sheet population. In the region of overlap, the proton density (*n*) and temperature (*T*) have opposite gradients (white arrowheads). Both wave types are excited outside the plasmasphere.

8. Discussion

The major finding of this study is that symmetric compressional waves were excited at 3.0 < L < 5.5 during a geomagnetic storm. Figure 14 provides a graphical description of this finding. The symmetric waves (wavy green curve) occur outside the plasmasphere (shaded blue) in the region where the energized in situ population (energies <5 keV, shaded green) overlaps the inner edge of the eastward-drifting energetic ions (energies >5 keV, extending to >100 keV, shaded orange) that are injected from the plasma sheet. The in situ population has a higher density (*n*) and a lower temperature (*T*) than the injected energetic ions, leading to an inward density gradient and an outward temperature gradient when both populations are included in moment calculations. These gradients, which favor DCI, are colocated with the waves at $L \simeq 3.0-5.0$ in the example shown in Figure 9. We cannot determine the MLT extent of the symmetric waves because the MLT range covered by RBSP is limited in the present study. Therefore, the MLT span of the symmetric waves illustrated in Figure 14 could be very different from reality.

An important factor to be considered for the symmetric waves is that they occur in the region of $\beta < 1$. This is because the magnetic field is strong (B > 250 nT) at L < 5 so that the magnetic field pressure is higher than the ion pressure even when the high-energy plasma sheet population and the energized in situ population are simultaneously present. Note that the latter population does not contribute much to β because it appears at energies lower than the core of the pressure-carrying ring current population (Williams, 1981). The moderate β value makes the DMI an unlikely mechanism for the symmetric waves. This warrants a different theoretical approach to the symmetric waves compared to that for compressional Pc5 waves observed at L > 5. In addition to DCI, DRI is a possible mechanism to excite the symmetric waves.

The symmetric waves are distinct from antisymmetric poloidal waves (wavy red curve) routinely detected at L > 5 primarily on the afternoon side (Anderson et al., 1990; Kokubun, 1985) often with a very strong magnetic field compressional component. The energy source of the antisymmetric waves is believed to be the injected energetic ions. The ions occupy a region extending to L > 7, explaining why the waves are excited beyond geosynchronous orbit. For the antisymmetric waves, there is an established theoretical framework, which incorporates

ring current ion pressure, coupling between Alfvén and compressional modes, and effects of bouncing particles (Chen & Hasegawa, 1991; Cheng & Lin, 1987; Southwood, 1976). The free energy sources for the waves include a bump-on-tail energy distribution (Takahashi, Oimatsu, et al., 2018) and a radial gradient of the phase space density (Oimatsu et al., 2018). Quantitative tests of the generation mechanisms are possible for the antisymmetric waves because observations provide not only the relevant background plasma parameters but also the phase and amplitude relationship between the wave fields and the perturbed plasma pressure and ion fluxes (Kokubun et al., 1989; Kremser et al., 1981). In addition, it is possible to determine m through interspacecraft phase delay analysis (Takahashi et al., 1985) or using ion finite Larmor radius sounding techniques (Lin et al., 1988; Min et al., 2017). In future studies, we will look for symmetric waves to which we can apply these techniques to better specify the wave properties. Knowledge of m is also necessary to understand the response of electrons to the waves.

Whether the energized in situ population is indeed a prerequisite for the symmetric waves needs to be determined by analyzing more events. The population was present on the few RBSP equatorial passes with symmetric waves during the selected geomagnetic storm. Whether or not this is a coincidence can be determined by a statistical analysis of stormtime ULF waves encountered by RBSP over the mission period of 7 years. We note in Figure 2 that the first two symmetric wave events highlighted (orbits 2920 and 2922) occurred a few hours after very strong substorm activity (AL $\leq -1,000$ nT) and that the third event also occurred ~8 hr after another (less intense) substorm. In addition, the wave power in those three symmetric ULF wave cases seems to correlate to the length of time passed since the previous substorm. By contrast, there is no such substorm activity before the antisymmetric waves observed on the fourth highlighted event in Figure 2. The appearance of the in situ population and the symmetric waves could be related to the history of substorm activity. The high-energy plasma sheet population is obviously caused by ion injection during substorms, and plasma waves excited by this population might play a role in the formation of the in situ population.

9. Conclusions

We have studied ULF waves in the inner magnetosphere excited during a moderate geomagnetic storm. Analysis of RBSP and ground magnetometer data revealed the following properties of the waves:

- 1. The waves are excited at L = 3-5.5 outside the plasmasphere.
- 2. At the magnetic equator, the waves produce strong perturbations in E_{ν} , E_{ϕ} , and B_{μ} but not in B_{ν} or B_{ϕ} , consistent with a symmetric mode structure about the equator.
- 3. The waves are detected in the region of ion flux enhancement at energies lower than \sim 5 keV.
- 4. Electron fluxes at energies from ~30 keV to ~1 MeV oscillated in phase with B_{μ} regardless of the pitch angle.

We also discussed possible excitation mechanisms for the symmetric waves:

- 1. The threshold for DMI is not met.
- 2. The phase space density of protons shows an inward gradient at some energy, which could drive DRI if the waves have a very high azimuthal wavenumber.
- 3. Proton bulk parameters show properties consistent with DCI.

Data Availability Statement

The data used in this study are publicly available from the following sources: NASA Goddard Space Flight Center (GSFC) Space Physics Data Facility Coordinated Data Analysis Web (https://cdaweb.gsfc.nasa.gov/index.html) for RBSP data; NASA/GSFC Space Physics Data Facility OMNIWeb Plus (https://omniweb.gsfc.nasa.gov/) for Solar wind OMNI data; Zenodo archive (https://doi.org/10.5281/zenodo.5639728) for EMMA data; and World Data Center of Geomagnetism, Kyoto (http://wdc.kugi.kyoto-u.ac.jp/) for geomagnetic activity indices.



Acknowledgments

K.T. was supported by NASA Grants NNX17AD34G and 80NSSC19K0259. R.E.D. was supported by NASA Grants 80NSSC20K1446 and 80NSSC21K0543. Work at The Aerospace Corporation was supported by Energetic Particle, Composition, and Thermal Plasma (RBSP-ECT) investigation funded under NASA's Prime contract no. NAS5-01072. The RBSPICE instrument was supported by JHU/APL subcontract 937836 to the New Jersey Institute of Technology under NASA Prime contract NAS5-01072. The authors thank J. Manwailer for providing information on RBSPICE data products. The authors thank the Finnish Meteorological Institute (FMI), the University of Oulu (Finland). the Institute of Geophysics of the Polish Academy of Sciences (IGF-PAS), the Mining and Geological Survey of Hungary (MBFSZ), and the University of L'Aquila for contributing to EMMA.

References

- Anderson, B. J., Engebretson, M. J., Rounds, S. P., Zanetti, L. J., & Potemra, T. A. (1990). A statistical study of Pc 3–5 pulsations observed by the AMPTE/CCE magnetic fields experiment, 1. Occurrence distributions. *Journal of Geophysical Research*, 95(A7), 10495–10523. https:// doi.org/10.1029/JA095iA07p10495
- Barfield, J. N., & McPherron, R. L. (1972). Statistical characteristics of storm-associated Pc 5 micropulsations observed at the synchronous equatorial orbit. *Journal of Geophysical Research*, 77(25), 4720–4733. https://doi.org/10.1029/JA077i025p04720
- Berube, D., Moldwin, M. B., & Ahn, M. (2006). Computing magnetospheric mass density from field line resonances in a realistic magnetic field geometry. *Journal of Geophysical Research*, 111(A8), A08206. https://doi.org/10.1029/2005ja011450
 - Blake, J. B., Carranza, P. A., Claudepierre, S. G., Clemmons, J. H., Crain, W. R., Dotan, Y., et al. (2013). The Magnetic Electron Ion Spectrometer (MagEIS) instruments aboard the Radiation Belt Storm Probes (RBSP) spacecraft. Space Science Reviews, 179(1-4), 383–421. https://doi.org/10.1007/s11214-013-9991-8
 - Brekke, A., Feder, T., & Berger, S. (1987). Pc4 giant pulsations recorded in Tromso, 1929–1985. Journal of Atmospheric and Terrestrial Physics, 49(10), 1027–1032. https://doi.org/10.1016/0021-9169(87)90109-7
 - Chelpanov, M. A., Mager, O. V., Mager, P. N., Klimushkin, D. Y., & Berngardt, O. I. (2018). Properties of frequency distribution of Pc5-range pulsations observed with the Ekaterinburg decameter radar in the nightside ionosphere. *Journal of Atmospheric and Solar-Terrestrial Physics*, 167, 177–183. https://doi.org/10.1016/j.jastp.2017.12.002
 - Chelpanov, M. A., Mager, P. N., Klimushkin, D. Y., Berngardt, O. I., & Mager, O. V. (2016). Experimental evidence of drift compressional waves in the magnetosphere: An Ekaterinburg coherent decameter radar case study. *Journal of Geophysical Research: Space Physics*, 121(2), 1315–1326. https://doi.org/10.1002/2015ja022155
 - Cheng, C. Z., & Lin, C. S. (1987). Eigenmode analysis of compressional waves in the magnetosphere. *Geophysical Research Letters*, 14(8), 884–887. https://doi.org/10.1029/GL014i008p00884
 - Chen, L., & Hasegawa, A. (1991). Kinetic theory of geomagnetic pulsations: 1. Internal excitations by energetic particles. *Journal of Geophysical Research: Space Physics*, 96(A2), 1503–1512. https://doi.org/10.1029/90ja02346
 - Claudepierre, S. G., Mann, I. R., Takahashi, K., Fennell, J. F., Hudson, M. K., Blake, J. B., et al. (2013). Van Allen Probes observation of localized drift resonance between poloidal mode ultra-low frequency waves and 60 keV electrons. *Geophysical Research Letters*, 40(17), 4491–4497. https://doi.org/10.1002/grl.50901
 - Constantinescu, O. D., Glassmeier, K. H., Plaschke, F., Auster, U., Angelopoulos, V., Baumjohann, W., et al. (2009). THEMIS observations of duskside compressional Pc5 waves. *Journal of Geophysical Research: Space Physics*, 114(A1A00C25). https://doi.org/10.1029/2008ja013519
 - Crabtree, C., & Chen, L. (2004). Finite gyroradius theory of drift compressional modes. *Geophysical Research Letters*, *31*, 17804. https://doi.org/10.1029/2004GL020660
 - Crabtree, C., Horton, W., Wong, H. V., & Van Dam, J. W. (2003). Bounce-averaged stability of compressional modes in geotail flux tubes. Journal of Geophysical Research: Space Physics, 108(A21084). https://doi.org/10.1029/2002ja009555
 - Cummings, W. D., O'Sullivan, R. J., & Coleman, P. J. (1969). Standing Alfvén waves in the magnetosphere. *Journal of Geophysical Research*, 74(3), 778–793. https://doi.org/10.1029/JA074i003p00778
 - Dai, L., Takahashi, K., Lysak, R., Wang, C., Wygant, J. R., Kletzing, C., et al. (2015). Storm time occurrence and spatial distribution of Pc4 poloidal ULF waves in the inner magnetosphere: A Van Allen Probes statistical study. *Journal of Geophysical Research: Space Physics*, 120(6), 4748–4762. https://doi.org/10.1002/2015ja021134
 - Dai, L., Takahashi, K., Wygant, J. R., Chen, L., Bonnell, J., Cattell, C. A., et al. (2013). Excitation of poloidal standing Alfvén waves through drift resonance wave-particle interaction. *Geophysical Research Letters*, 40(16), 4127–4132. https://doi.org/10.1002/grl.50800
- Denton, M. H., Reeves, G. D., Thomsen, M. F., Henderson, M. G., Friedel, R. H. W., Larsen, B., et al. (2016). The complex nature of storm-time ion dynamics: Transport and local acceleration. *Geophysical Research Letters*, 43(19), 10059–10067. https://doi.org/10.1002/2016gl070878
- Denton, R. E., Lessard, M. R., & Kistler, L. M. (2003). Radial localization of magnetospheric guided poloidal Pc 4–5 waves. Journal of Geophysical Research, 108(A3), 1105. https://doi.org/10.1029/2002ja009679
- Elkington, S. R., Hudson, M. K., & Chan, A. A. (1999). Acceleration of relativistic electrons via drift-resonant interaction with toroidal-mode Pc-5 ULF oscillations. *Geophysical Research Letters*, 26(21), 3273–3276. https://doi.org/10.1029/1999gl003659
- Eriksson, P. T. I., Blomberg, L. G., & Glassmeier, K. H. (2006). Cluster satellite observations of mHz pulsations in the dayside magnetosphere. Advances in Space Research, 38(8), 1730–1737. https://doi.org/10.1016/j.asr.2005.04.103
- Eriksson, P. T. I., Blomberg, L. G., Walker, A. D. M., & Glassmeier, K.-H. (2005). Poloidal ULF oscillations in the dayside magnetosphere: A cluster study. Annales Geophysicae, 23, 2679–2686. https://doi.org/10.5194/angeo-23-2679-2005
- Fälthammar, C.-G. (1965). Effects of time-dependent electric fields on geomagnetically trapped radiation. *Journal of Geophysical Research*, 70(11), 2503–2516. https://doi.org/10.1029/JZ070i011p02503
- Funsten, H. O., Skoug, R. M., Guthrie, A. A., MacDonald, E. A., Baldonado, J. R., Harper, R. W., et al. (2013). Helium, Oxygen, Proton, and Electron (HOPE) mass spectrometer for the radiation belt storm probes mission. *Space Science Reviews*, 179(1–4), 423–484. https://doi. org/10.1007/s11214-013-9968-7

Green, C. A. (1985). Giant pulsations in the plasmasphere. *Planetary and Space Science*, 33(10), 1155–1168. https://doi.org/10.1016/0032-0633(85)90073-x

- Green, J. C., & Kivelson, M. G. (2001). A tale of two theories: How the adiabatic response and ULF waves affect relativistic electrons. Journal of Geophysical Research: Space Physics, 106(A11), 25777–25791. https://doi.org/10.1029/2001ja000054
- Hamlin, D. A., Karplus, R., Vik, R. C., & Watson, K. M. (1961). Mirror and azimuthal drift frequencies for geomagnetically trapped particles. *Journal of Geophysical Research*, 66(1), 1–4. https://doi.org/10.1029/JZ066i001p00001

Hasegawa, A. (1969). Drift mirror instability in the magnetosphere. Physics of Fluids, 12(12), 2642-2650. https://doi.org/10.1063/1.1692407

- Hasegawa, A. (1971a). Drift-wave instabilities of a compressional mode in a high-β plasma. *Physical Review Letters*, 27(1), 11–14. https://doi.org/10.1103/PhysRevLett.27.11
- Hasegawa, A. (1971b). Drift-wave instability at the plasmapause. Journal of Geophysical Research, 76(22), 5361–5364. https://doi.org/10.1029/ JA076i022p05361
- Hilmer, R. V., Ginet, G. P., & Cayton, T. E. (2000). Enhancement of equatorial energetic electron fluxes near L= 4.2 as a result of high speed solar wind streams. Journal of Geophysical Research: Space Physics, 105(A10), 23311–23322. https://doi.org/10.1029/1999ja000380
- Hughes, W. J., & Southwood, D. J. (1976). The screening of micropulsation signals by the atmosphere and ionosphere. *Journal of Geophysical Research*, 81(19), 3234–3240. https://doi.org/10.1029/JA081i019p03234
- Hughes, W. J., Southwood, D. J., Mauk, B., McPherron, R. L., & Barfield, J. N. (1978). Alfvén waves generated by an inverted plasma energy distribution. *Nature*, 275(5675), 43–45. https://doi.org/10.1038/275043a0

- Kivelson, M. G., & Southwood, D. J. (1985). Charged particle behavior in low-frequency geomagnetic pulsations: 4. Compressional waves. Journal of Geophysical Research, 90(A2), 1486. https://doi.org/10.1029/JA090iA02p01486
- Kletzing, C. A., Kurth, W. S., Acuna, M., MacDowall, R. J., Torbert, R. B., Averkamp, T., et al. (2013). The Electric and Magnetic Field Instrument Suite and Integrated Science (EMFISIS) on RBSP. Space Science Reviews, 179(1–4), 127–181. https://doi.org/10.1007/s11214-013-9993-6
- Kokubun, S. (1985). Statistical characteristics of Pc5 waves at geostationary orbit. Journal of Geomagnetism and Geoelectricity, 37(8), 759–779. https://doi.org/10.5636/jgg.37.759
- Kokubun, S., Erickson, K. N., Fritz, T. A., & Mcpherron, R. L. (1989). Local time asymmetry of Pc-4–5 pulsations and associated particle modulations at synchronous orbit. *Journal of Geophysical Research: Space Physics*, 94(A6), 6607–6625. https://doi.org/10.1029/JA094iA06p06607
- Kremser, G., Korth, A., Fejer, J. A., Wilken, B., Gurevich, A. V., & Amata, E. (1981). Observations of quasi-periodic flux variations of energetic ions and electrons associated with Pc 5 geomagnetic pulsations. *Journal of Geophysical Research*, 86(A5), 3345. https://doi.org/10.1029/JA086iA05p03345
- Kurth, W. S., De Pascuale, S., Faden, J. B., Kletzing, C. A., Hospodarsky, G. B., Thaller, S., et al. (2015). Electron densities inferred from plasma wave spectra obtained by the Waves instrument on Van Allen Probes. *Journal of Geophysical Research: Space Physics*, 120(2), 904–914. https://doi.org/10.1002/2014JA020857
- Lanzerotti, L. J., Hasegawa, A., & Maclennan, C. G. (1969). Drift mirror instability in the magnetosphere: Particle and field oscillations and electron heating. *Journal of Geophysical Research*, 74(24), 5565–5578. https://doi.org/10.1029/JA074i024p05565
- Le, G., Chi, P. J., Strangeway, R. J., Russell, C. T., Slavin, J. A., Takahashi, K., et al. (2017). Global observations of magnetospheric high-m poloidal waves during the 22 June 2015 magnetic storm. Geophysical Research Letters, 44(8), 3456–3464. https://doi.org/10.1002/2017GL073048
- Lichtenberger, J., Clilverd, M. A., Heilig, B., Vellante, M., Manninen, J., Rodger, C. J., et al. (2013). The plasmasphere during a space weather event: First results from the PLASMON project. *Journal of Space Weather and Space Climate*, 3, A23. https://doi.org/10.1051/swsc/2013045
- Lin, N., McPherron, R. L., Kivelson, M. G., & Williams, D. J. (1988). An unambiguous determination of the propagation of a compressional Pc 5 wave. *Journal of Geophysical Research*, 93(A6), 5601. https://doi.org/10.1029/JA093iA06p05601
- Liu, W., Cao, J. B., Li, X., Sarris, T. E., Zong, Q. G., Hartinger, M., et al. (2013). Poloidal ULF wave observed in the plasmasphere boundary layer. *Journal of Geophysical Research: Space Physics*, 118(7), 4298–4307. https://doi.org/10.1002/jgra.50427
- Liu, Z. Y., Zong, Q. G., Zhou, X. Z., Zhu, Y. F., & Gu, S. J. (2020). Pitch angle structures of ring current ions induced by evolving poloidal ultra?low frequency waves. *Geophysical Research Letters*, 47(4). https://doi.org/10.1029/2020g1087203
- Mager, O. V., Chelpanov, M. A., Mager, P. N., Klimushkin, D. Y., & Berngardt, O. I. (2019). Conjugate ionosphere-magnetosphere observations of a sub-Alfvénic compressional intermediate-m wave: A case study using EKB radar and Van Allen Probes. Journal of Geophysical Research: Space Physics, 124(5), 3276–3290. https://doi.org/10.1029/2019ja026541
- Mager, P. N., Klimushkin, D. Y., & Kostarev, D. V. (2013). Drift-compressional modes generated by inverted plasma distributions in the magnetosphere. Journal of Geophysical Research: Space Physics, 118(8), 4915–4923. https://doi.org/10.1002/jgra.50471
- Mann, I. R., Lee, E. A., Claudepierre, S. G., Fennell, J. F., Degeling, A., Rae, I. J., et al. (2013). Discovery of the action of a geophysical synchrotron in the Earth's Van Allen radiation belts. *Nature Communications*, 4. https://doi.org/10.1038/ncomms3795
- Mauk, B. H., Fox, N. J., Kanekal, S. G., Kessel, R. L., Sibeck, D. G., & Ukhorskiy, A. (2012). Science objectives and rationale for the radiation belt storm probes mission. Space Science Reviews, 179(1–4), 3–27. https://doi.org/10.1007/s11214-012-9908-y
- Min, K., Takahashi, K., Ukhorskiy, A. Y., Manweiler, J. W., Spence, H. E., Singer, H. J., et al. (2017). Second harmonic poloidal waves observed by Van Allen Probes in the dusk-midnight sector. *Journal of Geophysical Research: Space Physics*, 122(3), 3013–3039. https://doi. org/10.1002/2016ja023770
- Mitchell, D. G., Lanzerotti, L. J., Kim, C. K., Stokes, M., Ho, G., Cooper, S., et al. (2013). Radiation Belt Storm Probes Ion Composition Experiment (RBSPICE). Space Science Reviews, 179(1–4), 263–308. https://doi.org/10.1007/s11214-013-9965-x
- Motoba, T., Takahashi, K., Rodriguez, J. V., & Russell, C. T. (2015). Giant pulsations on the afternoonside: Geostationary satellite and ground observations. Journal of Geophysical Research: Space Physics, 120, 8350–8367. https://doi.org/10.1002/2015ja021592
- Ng, P. H., Patel, V. L., & Chen, S. (1984). Drift compressional instability in the magnetosphere. *Journal of Geophysical Research*, 89(A12), 10763. https://doi.org/10.1029/ja089ia12p10763
- Nishida, A., Mukai, T., Yamamoto, T., Saito, Y., Kokubun, S., & Lepping, R. P. (1997). Traversal of the nightside magnetosphere at 10 to 15 Re during northward IMF. *Geophysical Research Letters*, 24(8), 939–942. https://doi.org/10.1029/97g100219
- Oimatsu, S., Nosé, M., Takahashi, K., Yamamoto, K., Keika, K., Kletzing, C. A., et al. (2018). Van Allen Probes observations of drift-bounce resonance and energy transfer between energetic ring current protons and poloidal Pc4 wave. *Journal of Geophysical Research: Space Physics*, 123, 3421–3435. https://doi.org/10.1029/2017ja025087
- Radoski, H. R. (1967). Highly asymmetric MHD resonances: The guided poloidal mode. Journal of Geophysical Research, 72(15), 4026–4027. https://doi.org/10.1029/JZ072i015p04026
- Radoski, H. R., & Carovillano, R. L. (1966). Axisymmetric plasmasphere resonances Toroidal mode. *Physics of Fluids*, 9(2), 285–291. https://doi.org/10.1063/1.1761671
- Rae, J. I., Murphy, K. R., Watt, C. E. J., Halford, A. J., Mann, I. R., Ozeke, L. G., et al. (2018). The role of localized compressional ultra-low frequency waves in energetic electron precipitation. *Journal of Geophysical Research: Space Physics*, 123, 1900–1914. https://doi. org/10.1002/2017ja024674
- Rosenbluth, M. N. (1981). Magnetic trapped-particle modes. *Physical Review Letters*, 46(23), 1525–1528. https://doi.org/10.1103/ PhysRevLett.46.1525
- Rubtsov, A. V., Agapitov, O. V., Mager, P. N., Klimushkin, D. Y., Mager, O. V., Mozer, F. S., et al. (2018). Drift resonance of compressional ULF waves and substorm-injected protons from multipoint THEMIS measurements. *Journal of Geophysical Research: Space Physics*, 123(11), 9406–9419. https://doi.org/10.1029/2018ja025985
- Shue, J. H., Song, P., Russell, C. T., Steinberg, J. T., Chao, J. K., Zastenker, G., et al. (1998). Magnetopause location under extreme solar wind conditions. *Journal of Geophysical Research: Space Physics*, 103(A8), 17691–17700. https://doi.org/10.1029/98ja01103
- Sonnerup, B. U. Ö., Cahill, L. J., & Davis, L. R. (1969). Resonant vibration of the magnetosphere observed from Explorer 26. Journal of Geophysical Research, 74(9), 2276–2288. https://doi.org/10.1029/JA074i009p02276
- Soto-Chavez, A. R., Lanzerotti, L. J., Manweiler, J. W., Gerrard, A., Cohen, R., Xia, Z., et al. (2019). Observational evidence of the drift-mirror plasma instability in Earth's inner magnetosphere. *Physics of Plasmas*, 26(4), 042110. https://doi.org/10.1063/1.5083629
- Southwood, D. J. (1973). The behaviour of ULF waves and particles in the magnetosphere. *Planetary and Space Science*, 21(1), 53–65. https://doi.org/10.1016/0032-0633(73)90019-6
- Southwood, D. J. (1976). A general approach to low-frequency instability in the ring current plasma. *Journal of Geophysical Research*, 81(19), 3340–3348. https://doi.org/10.1029/JA081i019p03340



- Southwood, D. J., & Kivelson, M. G. (1981). Charged particle behavior in low-frequency geomagnetic pulsations 1. Transverse waves. Journal of Geophysical Research, 86(A7), 5643. https://doi.org/10.1029/JA086iA07p05643
- Southwood, D. J., & Kivelson, M. G. (1982). Charged particle behavior in low-frequency geomagnetic pulsations, 2. Graphical approach. Journal of Geophysical Research: Space Physics, 87(A3), 1707–1710. https://doi.org/10.1029/JA087iA03p01707
- Southwood, D. J., & Kivelson, M. G. (1993). Mirror instability: 1. Physical mechanism of linear instability. Journal of Geophysical Research, 98(A6), 9181–9187. https://doi.org/10.1029/92ja02837
- Takahashi, K., Cheng, C. Z., McEntire, R. W., & Kistler, L. M. (1990). Observation and theory of Pc 5 waves with harmonically related transverse and compressional components. *Journal of Geophysical Research*, 95(A2), 977. https://doi.org/10.1029/JA095iA02p00977
- Takahashi, K., Claudepierre, S. G., Rankin, R., Mann, I., & Smith, C. W. (2018). Van Allen Probes observation of a fundamental poloidal standing Alfvén wave event related to giant pulsations. *Journal of Geophysical Research: Space Physics*, 123(6), 4574–4593. https://doi.org/10.1029/2017ja025139
- Takahashi, K., & Denton, R. E. (2021). Nodal structure of toroidal standing Alfvén waves and its implication for field line mass density distribution. Journal of Geophysical Research: Space Physics, 126, e2020JA028981. https://doi.org/10.1029/2020JA028981
- Takahashi, K., Denton, R. E., Anderson, R. R., & Hughes, W. J. (2006). Mass density inferred from toroidal wave frequencies and its comparison to electron density. *Journal of Geophysical Research: Space Physics*, 111(A1), A01201. https://doi.org/10.1029/2005ja011286
- Takahashi, K., Fennell, J. F., Amata, E., & Higbie, P. R. (1987). Field-aligned structure of the storm time Pc 5 wave of November 14–15, 1979. Journal of Geophysical Research, 92(A6), 5857. https://doi.org/10.1029/JA092iA06p05857
- Takahashi, K., Glassmeier, K. H., Angelopoulos, V., Bonnell, J., Nishimura, Y., Singer, H. J., et al. (2011). Multisatellite observations of a giant pulsation event. Journal of Geophysical Research: Space Physics, 116(A11), A11223. https://doi.org/10.1029/2011ja016955
- Takahashi, K., Hartinger, M. D., Angelopoulos, V., Glassmeier, K.-H., & Singer, H. J. (2013). Multispacecraft observations of fundamental poloidal waves without ground magnetic signatures. *Journal of Geophysical Research: Space Physics*, 118(7), 4319–4334. https://doi.org/10.1002/ jgra.50405
- Takahashi, K., Higbie, P. R., & Baker, D. N. (1985). Azimuthal propagation and frequency characteristic of compressional Pc 5 waves observed at geostationary orbit. *Journal of Geophysical Research*, *90*(A2), 1473. https://doi.org/10.1029/JA090iA02p01473
- Takahashi, K., Oimatsu, S., Nosé, M., Min, K., Claudepierre, S. G., Chan, A., et al. (2018). Van Allen Probes observations of second harmonic poloidal standing Alfvén waves. Journal of Geophysical Research: Space Physics, 123, 611–637. https://doi.org/10.1002/2017ja024869
- Thompson, S. M., & Kivelson, M. G. (2001). New evidence for the origin of giant pulsations. Journal of Geophysical Research: Space Physics, 106(A10), 21237–21253. https://doi.org/10.1029/2001ja000026
- Tsyganenko, N. A. (1989). A magnetospheric magnetic field model with a warped tail current sheet. *Planetary and Space Science*, 37(1), 5–20. https://doi.org/10.1016/0032-0633(89)90066-4
- Tsyganenko, N. A. (2002). A model of the near magnetosphere with a dawn-dusk asymmetry 1. Mathematical structure. Journal of Geophysical Research: Space Physics, 107(A8), SMP12-1–SMP12-15. https://doi.org/10.1029/2001ja000219
- Tsyganenko, N. A., & Sitnov, M. I. (2005). Modeling the dynamics of the inner magnetosphere during strong geomagnetic storms. Journal of Geophysical Research: Space Physics, 110(A3), A03208. https://doi.org/10.1029/2004ja010798
- Ukhorskiy, A. Y., Sitnov, M. I., Takahashi, K., & Anderson, B. J. (2009). Radial transport of radiation belt electrons due to stormtime Pc5 waves. Annales Geophysicae, 27(5), 2173–2181. https://doi.org/10.5194/angeo-27-2173-2009
- Vaivads, A., Baumjohann, W., Haerendel, G., Nakamura, R., Kucharek, H., Klecker, B., et al. (2001). Compressional Pc5 type pulsations in the morning plasma sheet. *Annales Geophysicae*, 19, 311–320. https://doi.org/10.5194/angeo-19-311-2001
- Waters, C. L., Menk, F. W., & Fraser, B. J. (1991). The resonance structure of low latitude Pc3 geomagnetic-pulsations. *Geophysical Research Letters*, 18(12), 2293–2296. https://doi.org/10.1029/91gl02550
- Williams, D. (1981). Ring current composition and sources: An update. *Planetary and Space Science*, 29(11), 1195–1203. https://doi.org/10.1016/0032-0633(81)90124-0
- Wright, D. M., & Yeoman, T. K. (1999). High-latitude HF Doppler observations of ULF waves: 2. Waves with small spatial scale sizes. Annales Geophysicae, 17, 868–876. https://doi.org/10.1007/s00585-999-0868-9
- Wygant, J. R., Bonnell, J. W., Goetz, K., Ergun, R. E., Mozer, F. S., Bale, S. D., et al. (2013). The electric field and waves instruments on the radiation belt storm probes mission. *Space Science Reviews*, 179(1–4), 183–220. https://doi.org/10.1007/s11214-013-0013-7
- Yamamoto, K., Nosé, M., Kasahara, S., Yokota, S., Keika, K., Matsuoka, A., et al. (2018). Giant pulsations excited by a steep earthward gradient of proton phase space density: Arase observation. *Geophysical Research Letters*, 45, 6773–6781. https://doi.org/10.1029/2018gl078293
- Yamamoto, K., Nose, M., Keika, K., Hartley, D. P., Smith, C. W., MacDowall, R. J., et al. (2019). Eastward propagating second harmonic poloidal waves triggered by temporary outward gradient of proton phase space density: Van Allen Probe A observation. *Journal of Geophysical Research: Space Physics*, 124(12), 9904–9923. https://doi.org/10.1029/2019ja027158
- Yeoman, T. K., Wright, D. M., Chapman, P. J., & Stockton-Chalk, A. B. (2000). High-latitude observations of ULF waves with large azimuthal wavenumbers. *Journal of Geophysical Research: Space Physics*, 105(A3), 5453–5462. https://doi.org/10.1029/1999ja005081
- Zhu, X., & Kivelson, M. G. (1994). Compressional ULF waves in the outer magnetosphere: 2. A case-study of Pc-5 type wave activity. Journal of Geophysical Research: Space Physics, 99(A1), 241–252. https://doi.org/10.1029/93ja02106
- Zhu, X., & Kivelson, M. G. (1991). Compressional ULF waves in the outer magnetosphere: 1. Statistical study. Journal of Geophysical Research, 96(A11), 19451. https://doi.org/10.1029/91ja01860
- Zong, Q., Rankin, R., & Zhou, X. (2017). The interaction of ultra-low-frequency pc3–5 waves with charged particles in Earth's magnetosphere. *Reviews of Modern Plasma Physics*, 1(1). https://doi.org/10.1007/s41614-017-0011-4